INTRODUCTION TO

# Paralegalism

PERSPECTIVES, PROBLEMS, AND SKILLS

EIGHTH EDITION



WILLIAM P. STATSKY

This is an electronic version of the print textbook. Due to electronic rights restrictions, some third party content may be suppressed. Editorial review has deemed that any suppressed content does not materially affect the overall learning experience. The publisher reserves the right to remove content from this title at any time if subsequent rights restrictions require it. For valuable information on pricing, previous editions, changes to current editions, and alternate formats, please visit <a href="www.cengage.com/highered">www.cengage.com/highered</a> to search by ISBN, author, title, or keyword for materials in your areas of interest.

Important notice: Media content referenced within the product description or the product text may not be available in the eBook version.

# INTRODUCTION TO 1 1.

# Paralegalism

PERSPECTIVES, PROBLEMS, AND SKILLS

EIGHTH EDITION

## **CENGAGE Learning**













#### Options.

Over 300 products in every area of the law: textbooks, digital resources, online courses, and more – helping you succeed in class and on the job.

#### Support.

We offer unparalleled, practical support: robust instructor and student supplements to ensure the best learning experience, custom publishing to meet your unique needs, and other benefits such as our Student Achievement Award. And our sales representatives are always ready to provide you with dependable service.

#### Feedback.

As always, we want to hear from you! Your feedback is our best resource for improving the quality of our products. Contact your sales representative or write us at the address below if you have any comments about our materials or if you have a product proposal.

Accounting and Financials for the Law Office • Administrative Law • Alternative Dispute Resolution • Bankruptcy Business Organizations/Corporations • Careers and Employment • Civil Litigation and Procedure • CP Exam Preparation • Computer Applications in the Law Office • Constitutional Law • Contract Law • Criminal Law and Procedure • Document Preparation • Elder Law • Employment Law • Environmental Law • Ethics • Evidence Law • Family Law • Health Care Law • Immigration Law • Intellectual Property • Internships Interviewing and Investigation • Introduction to Law • Introduction to Paralegalism • Juvenile Law • Law Office Management • Law Office Procedures • Legal Research, Writing, and Analysis • Legal Terminology • Legal Transcription • Media and Entertainment Law • Medical Malpractice Law Product Liability • Real Estate Law • Reference Materials • Social Security • Torts and Personal Injury Law • Wills, Trusts, and Estate Administration • Workers' Compensation Law

CENGAGE Learning 5 Maxwell Drive Clifton Park, New York 12065-2919

For additional information, find us online at: www.cengage.com



#### INTRODUCTION TO

# Paralegalism

PERSPECTIVES, PROBLEMS, AND SKILLS

EIGHTH EDITION



WILLIAM P. STATSKY





#### Introduction to Paralegalism: Perspectives, Problems, and Skills, Eighth Edition William P. Statsky

Senior Vice President, General Manager for Skills and Global Product Management: Dawn Gerrain

Product Manager: Katie McGuire

Senior Director, Development/Global Product Management, Skills: Marah Bellegarde

Senior Product Development Manager:

Larry Main

Senior Content Developer: Melissa Riveglia Senior Product Assistant: Diane E. Chrysler Marketing Manager: Scott Chrysler

Senior Production Director: Wendy Troeger

Production Director: Andrew Crouth Senior Content Project Manager:

Betty L. Dickson

Art Director: Brenda Carmichael, Lumina

Datamatics, Inc.

Senior Technology Project Manager: Joe Pliss

Cover image(s): White Marble Ionic Column Scroll Pediment Close-Up

Credit: © PeskyMonkey

White Marble Ionic Column Scroll Pediment

Close-Up, Credit: © PeskyMonkey

© 2016, 2009, 2003, 1997, 1992, 1986, 1982, 1974 Cengage Learning

WCN: 01-100-101

ALL RIGHTS RESERVED. No part of this work covered by the copyright herein may be reproduced, transmitted, stored, or used in any form or by any means graphic, electronic, or mechanical, including but not limited to photocopying, recording, scanning, digitizing, taping, Web distribution, information networks, or information storage and retrieval systems, except as permitted under Section 107 or 108 of the 1976 United States Copyright Act, without the prior written permission of the publisher.

For product information and technology assistance, contact us at Cengage Learning Customer & Sales Support, 1-800-354-9706

For permission to use material from this text or product, submit all requests online at www.cengage.com/permissions.

Further permissions questions can be e-mailed to permissionrequest@cengage.com

Library of Congress Control Number: 2014957050

ISBN: 978-1-285-44905-0

#### **Cengage Learning**

20 Channel Center Street Boston, MA 02210 USA

Cengage Learning is a leading provider of customized learning solutions with office locations around the globe, including Singapore, the United Kingdom, Australia, Mexico, Brazil, and Japan. Locate your local office at:

www.cengage.com/global

Cengage Learning products are represented in Canada by Nelson Education, Ltd.

To learn more about Cengage Learning, visit www.cengage.com

Purchase any of our products at your local college store or at our preferred online store **www.cengagebrain.com** 

#### Notice to the Reader

Publisher does not warrant or guarantee any of the products described herein or perform any independent analysis in connection with any of the product information contained herein. Publisher does not assume, and expressly disclaims, any obligation to obtain and include information other than that provided to it by the manufacturer. The reader is expressly warned to consider and adopt all safety precautions that might be indicated by the activities described herein and to avoid all potential hazards. By following the instructions contained herein, the reader willingly assumes all risks in connection with such instructions. The reader is notified that this text is an educational tool, not a practice book. Since the law is in constant change, no rule or statement of law in this book should be relied upon for any service to the client. The reader should always refer to standard legal sources for the current rule or law. If legal advice or other expert assistance is required, the services of the appropriate professional should be sought. The publisher makes no representations or warranties of any kind, including but not limited to, the warranties of fitness for particular purpose or merchantability, nor are any such representations implied with respect to the material set forth herein, and the publisher takes no responsibility with respect to such material. The publisher shall not be liable for any special, consequential, or exemplary damages resulting, in whole or part, from the readers' use of, or reliance upon, this material.

Printed in the United States of America Print Number: 02 Print Year: 2016 For Patricia Farrell Statsky, whose wisdom, light, and love have sustained more than she knows



#### **BYTHE SAME AUTHOR**

For all publications by William Statsky, see statsky.blogspot.com

The California Paralegal: Essential Rules, Documents, and Resources. Clifton Park, N.Y.: Cengage Learning, 2008 (with S. Sandberg)

Case Analysis and Fundamentals of Legal Writing, 4th ed. St. Paul: West Group, 1995 (with J. Wernet)

Essentials of Paralegalism: Perspectives, Problems, and Skills, 5th ed. Clifton Park, N.Y.: Cengage Learning, 2010

Essentials of Torts, 3d ed. Clifton Park, N.Y.: Cengage Learning, 2012

Family Law: The Essentials, 3d ed. Clifton Park, N.Y.: Cengage learning, 2015

Family Law, 6th ed. Clifton Park, N.Y.: Cengage Learning, 2013

The Florida Paralegal: Essential Rules, Documents, and Resources. Clifton Park, N.Y.: Cengage Learning, 2009 (with B. Diotalevi & P. Linquist)

Inmate Involvement in Prison Legal Services: Roles and Training Options for the Inmate as Paralegal. Chicago: American Bar Association, Commission on Correctional Facilities and Services, 1974

Legal Desk Reference. St. Paul: West Group, 1990 (with B. Hussey, M. Diamond, & R. Nakamura)

The Legal Paraprofessional as Advocate and Assistant: Training Concepts and Materials. New York: Center on Social Welfare Policy and Law, 1971 (with P. Lang)

Legal Research and Writing: Some Starting Points, 5th ed. Clifton Park, N.Y.: Cengage Learning, 1999

Legal Thesaurus/Dictionary: A Resource for the Writer and Computer Researcher. St. Paul: West Group, 1985

Legislative Analysis and Drafting, 2d ed. St. Paul: West Group, 1984

The New York Paralegal: Essential Rules, Documents, and Resources. Clifton Park, N.Y.: Cengage Learning, 2009 (with R. Sarachan)

The Ohio Paralegal: Essential Rules, Documents, and Resources. Clifton Park, N.Y.: Cengage Learning, 2008 (with K. Reed & B. Moore)

Paralegal Employment: Facts and Strategies for the 1990s, 2d ed. St. Paul: West Group, 1993

Paralegal Ethics and Regulation, 2d ed. St. Paul: West Group, 1993

The Pennsylvania Paralegal: Essential Rules, Documents, and Resources. Clifton Park, N.Y.: Cengage Learning, 2009 (with J. DeLeo & J. Geis)

The Texas Paralegal: Essential Rules, Documents, and Resources. Clifton Park, N.Y.: Cengage Learning, 2009 (with L. Crossett)

Torts: Personal Injury Litigation, 5th ed. Clifton Park, N.Y.: Cengage Learning, 2011

Rights of the Imprisoned: Cases, Materials, and Directions. Indianapolis, Ind.: Bobbs-Merrill Company, 1974 (with R. Singer)

What Have Paralegals Done? A Dictionary of Functions. Washington, D.C.: National Paralegal Institute, 1973

# **CONTENTS IN BRIEF**

Preface • xiii	14 Introduction to Law Office Administration • 589			
How to Study Law in the Classroom and on the Job • xvii	Informal and Formal Administrative Advocacy • 618			
<ul> <li>PART I</li> <li>THE PARALEGAL IN THE LEGAL SYSTEM 1</li> <li>I Introduction to the Paralegal Career • 3</li> <li>Paralegal Employment • 34</li> <li>On-the-Job Realities:     The Assertive Paralegal • 127</li> <li>The Regulation of Paralegals • 155</li> </ul>	Appendix A Books, Articles, and Social Media on Paralegalism 634  Appendix B Paralegal, Bar, and Related Associations 635  Appendix C Federal Government Organization Chart 639  Appendix D State Survey: Opinions, Rules, Statutes, and Reports on Paralegal			
<ul> <li>5 Attorney Ethics and Paralegal Ethics • 209</li> <li>6 Introduction to the Legal System • 285</li> <li>PART II</li> </ul>	Ethics and Related Topics 640  Appendix E Ethical Codes of NALA,  NFPA, and NALS 731  Appendix F Paralegal Blogs 734  Appendix G Starting a Freelance			
<ul> <li>THE SKILLS OF A PARALEGAL 309</li> <li>Introduction to Legal Analysis • 311</li> <li>Legal Interviewing • 363</li> <li>Investigation in a Law Office • 401</li> <li>Litigation Assistantship • 441</li> <li>Legal Research • 490</li> </ul>	Appendix J State Government Paralegal Jobs 743 Appendix J State Resources 747 Appendix K News Stories, War Stories, and Parting Shots 755			
<ul><li>Legal Writing • 535</li><li>Computers in a Law Office • 561</li></ul>	Glossary • 761 Index • 795			



# **CONTENTS**

Preface • xiii

ref	face • xiii		Volunteering as a Way to Gain Legal	
Hov	w to Study Law in the Classroom		Experience • 119	
	and on the Job • xvii		Summary of Job Strategies • 120	
	Classroom Learning • xviii		Your Second Job • 121	
	On-the-Job Learning: The Art of Being	3	On-the-Job Realities:	
	Supervised • xxviii		The Assertive Paralegal	127
$\mathbf{D}_{i}$	ART I		The Ideal • 127	
			Communication • 129	
ГН	IE PARALEGAL IN THE LEGAL SYSTEM I		The Multigenerational Office • 130 Expectations • 132	
1	Introduction to the Paralegal Career 3		The Environment of a Law Office • 133	
	Launching Your Career • 3		Office Politics • 133	
	Do You Know What Time It Is? • 4		Self-assessment and Employer Evaluations • 135	
	The Scope of Your Legal Education • 6		Working with Attorneys • 139	
	Major Players • 6		Assertive/Nonassertive/Aggressive Behavior • 14	0
	Job Titles • 8		The Skill of Listening • 142	
	Definitions of a Paralegal • 13 Paralegal Fees • 15		Keeping a Diary or Journal and a Career  Development File • 144	
	Career Ladders in Large Law Offices • 20		Paralegal CLE • 145	
	Paralegal Salaries • 21		Responsibilities of Supervisors • 145	
	Factors Influencing the Growth of Paralegalism • 22		Relationship with Secretaries • 146	
	"Oh, You're Studying Law?" • 27		Sexual Harassment • 147	
	Your Online Reputation • 27		Working with Systems • 149	
	Conclusion • 28	4	The Regulation of Paralegals	155
2	Paralegal Employment 34		Kinds of Regulation • 155	
	The Job Market • 34		Defining the Practice of Law • 157	
	Where Paralegals Work • 35		Roles for Paralegals in Court Proceedings • 170	
	Paralegal Specialties: A Dictionary of Functions • 44		Roles for Paralegals in Administrative Proceedings • 17	74
	Finding a Job • 78		Licensing • 180	
	Using Social Media in the Job Search • 88		Reforms in the Practice of Law:	
	Résumé • 93		Attempts to Close the Justice Gap • 189	
	Cover Letter • 102		Certification • 191	
	Writing Samples • 104		Fair Labor Standards Act (FLSA) • 194	
	Job Interview • 105		Tort Liability • 197	
	Job-Search Notebook (Jsn) • 113		Malpractice Insurance • 200	Ċ

5	Attorney Ethics and Paralegal Ethics	209	8	Legal Interviewing	363		
	Ethics in the Real World • 210			Introduction • 364			
	Enforcing Ethics • 212			Client Relations in an Era of Lawyer Bashing • 3	364		
	Paralegal Codes • 217			Hiring the Firm • 365			
	Ethics Rules: Introduction • 218			Context of Interviewing in Litigation • 367			
	Formation of the Attorney-Client Relationship	• 219		Format of an Intake Memo • 368			
	Limited-Scope Representation • 220			What Facts do You Seek? Guides to the			
	Competence • 221			Formulation of Questions • 370			
	Diligence/Unwarranted Delay • 224			What Does the Client Want? • 376			
	Fees • 225			Assessing Your Own Bias • 377			
	Misconduct by Attorneys • 231			Communication Skills: Analysis of an Interview	• 378		
	Obstructing Access to Evidence • 233			Improving Your Interviewing Skills • 392			
	Crime or Fraud by Clients • 233			The "Difficult" Client • 392			
	Frivolous Positions and Harassment • 234						
	Safekeeping Property • 234		9	Investigation in a Law Office	40 I		
	False Evidence and Statements; Failure to			Paralegal Roles in Investigation • 402			
	Disclose • 235			Nature and Context of Investigation • 403			
	Confidentiality • 236			Fact Analysis: Organizing the Options • 406			
	Conflict of Interest • 243			Distortions in Investigation • 407			
	Communication with the Other Side • 257			Sources of Evidence/Sources of Leads • 409			
	Solicitation • 257			Gaining Access to Records • 411			
	Advertising • 259			Evaluating Evidence • 412			
	Payment for Referrals or Recommendations • 1	260		Interviewing Witnesses • 413			
	Reporting Professional Misconduct • 260	200		Special Investigative Problems • 416			
	Pro Bono Services • 263			License to Investigate? • 423			
	Withdrawal • 263			Basics of Evidence Law • 423			
	Appearance of Impropriety • 264			Evidence Log • 429			
	Unauthorized Practice of Law (UPL) • 265			Taking a Witness Statement • 429			
	Paralegals • 265			Settlement Work-Up • 431			
	Your Ethics or Your Job? • 276			Ethics and Investigation • 434			
	10a. 2ames 6. 10a. job. 276			-	441		
6	Introduction to the Legal System	285	10	Litigation Assistantship	441		
	Our Legal System • 285			Introduction • 441			
	Introduction to Law • 287			Overview of Civil and Criminal Litigation • 441			
	Judicial Branch • 293			Alternative Dispute Resolution (ADR) • 465			
	Executive Branch • 298			Litigation Assistant; Pretrial Stage • 467			
	Legislative Branch • 300			Litigation Assistant Trial Stage • 481			
				Litigation Assistant Appeal Stage • 483			
P	ART II			Litigation Assistant Enforcement Stage • 484			
Н	E SKILLS OF A PARALEGAL	309	- 11	Legal Research	490		
				Introduction • 491			
7	Introduction to Legal Analysis	311		What Is a Legal Research Expert? • 491			
	Legal Analysis Skills: Introduction • 311			Perseverance and Legal Research • 492			
	Legal Analysis Skill: Element Identification • 313	3		Terminology of Legal Research • 493			
	Legal Analysis Skills: Issue Spotting and	-		Legal Research on the Internet • 495			
	Phrasing • 318			Finding Traditional Law Libraries • 498			
	Legal Analysis Skills: Factor Analysis • 330			Ask an Online Librarian • 499			
	Memorandum of Law • 331			CARTWHEEL • 500			
	Introduction to Court Opinions • 340			Kinds of Authority • 503			
	Legal Analysis Skills: Briefing an Opinion • 346			Citation • 506			
	Legal Analysis Skills: Applying an Opinion • 351			Cite Checking • 513			
	regai viigisis skiiis. Abbisiila aii Ohiiiioii , 331			CICC CHECKING . 313			

	The First Level of Legal Research • 515		Timekeepi	ing • 598		
	Finding Case Law • 518		Categorie	s of Fees • 603		
	Finding Statutes • 520		Billing • 6	006		
	Finding Constitutional Law • 523		Client Tru	st Accounts • 608		
	Finding Administrative Law • 523		Administr	ative Reports • 609		
	Finding Court Rules (Rules of Court) • 525		Client File	Management • 611		
	Finding Local Law • 525		LE Julyania I	and Family I Advitation action		
	Finding International Law • 526			and Formal Administrative	410	
	Finding Secondary Authority (Subject-Matter		Advocacy		618	
	Research) • 526			Advocacy • 618		
	Westlaw and LexisNexis • 526	Informal Advocacy • 619				
	Research Self-Audit • 530		Procedural Due Process • 623			
2	Legal Writing	535	Formal Ad	vocacy • 625		
_		333	Appendix A	Books, Articles, and Social		
	Introduction • 535			Media on Paralegalism	634	
	Dangers of Copy and Paste • 537		Appendix B	Paralegal, Bar, and		
	Letter Writing • 538 Email Writing • 541			Related Associations	635	
	Instruments • 542		Appendix C	Federal Government	000	
	Pleadings • 542		, ppondix o	Organization Chart	639	
	Memorandum of Law • 542		Appendix D	State Survey: Opinions,	037	
	Appellate Brief • 542		Appendix D	Rules, Statutes, and		
	General Writing Guidelines • 546			Reports on Paralegal		
				1	640	
3	Computers in a Law Office	561	۸ س ما:ب د <b>ا</b>	Ethics and Related Topics	040	
	Introduction • 562		Appendix E	Ethical Codes of NALA,		
	What to Do on Your OwnNow • 562		A 1: E	NFPA, and NALS	731	
	Review of the Basics • 564		Appendix F	Paralegal Blogs	734	
	Introduction to the Internet • 575		Appendix G	Starting a Freelance		
	Internet Searching • 579			Paralegal Business	735	
	Security on the Internet • 582		Appendix H	Finding Law Offices	741	
	Ethical Issues • 583		Appendix I	State Government		
4	Introduction to Law Office Administration	589		Paralegal Jobs	743	
7		307	Appendix J	State Resources	747	
	Practice of Law in the Private Sector • 590		Appendix K	News Stories, War Stories,		
	Legal Administrator • 594 Paralegal Manager • 596			and Parting Shots	755	
	Outsourcing • 596					
	Virtual Law Practice • 597	Glossary • 76	Glossary • 761			
	Costs of Running a Law Office • 597		Index • 795			



### **PREFACE**

Seven editions ago — in 1974 — many were asking the question "What's a paralegal?" That day has long passed, although there is still a great deal of information that people need to have in order to appreciate the outstanding contribution paralegals have made in the delivery of legal services. This book seeks to provide that information and, at the same time, to introduce you to some of the fundamental skills needed to thrive in this still-developing career, a career whose members may one day outnumber attorneys in the traditional and untraditional law office.

It's a fascinating time to study law. So many aspects of our lives revolve around the making, interpretation, and application of laws. As integral members of the attorney's team, paralegals have a unique opportunity to view the legal system in operation as it wrestles with the legal issues of the day.

Back in the 1970s when the first edition of this book came out, Kathy Lowery, a Kansas City paralegal, was asked what impressed her most about the paralegal profession. She said that the two best things about the field were the two constants that were ever present: challenges and opportunities to "learn and grow." Well said about the 1970s and equally true as we continue into the twenty-first century.

#### **CHANGES IN THE EIGHTH EDITION**

In addition to the updated material in the book on employment, salaries, roles, ethics, and regulation, a number of particular changes in the eighth edition should be mentioned:

- New categories of assignments have been added, including assignments in critical analysis, the job search, and the core skills of writing, research, ethics, computers, and collaboration. All of the assignments have been placed at the end of each chapter.
- The role of social media is discussed throughout the chapters beginning with the admonition to control one's online reputation.

- Chapter 1 adds a discussion of the important difference between what paralegals need to understand the legal system and what they will need to perform specific tasks on the job.
- Chapter 2 adds extensive material on assertive networking and the use of LinkedIn, Twitter, and Facebook in the job search.
- The extensive coverage of private law firms and state government paralegal jobs has been moved from the end of Chapter 2 to the appendix material at the end of the book.
- Chapter 2 covers the major growth areas of law in the coming years.
- Chapter 2 includes a summary checklist of major job strategies, particularly in markets where competition for jobs is intense.
- Chapter 3 covers the dynamics of working in a multigenerational law office.
- Chapter 3 adds a section on the important and often overlooked skill of listening.
- Chapter 4 adds material on new proposals to end the justice gap, including expanded roles for the paralegal and other nonattorneys.
- Chapter 4 presents the major development of the Limited License Legal Technician in Washington and the favorable climate that exists in many states for recognizing document service providers (DSPs).
- Chapter 5 covers fraudulent "friending" and related ethical misconduct in the use of social media.
- Chapter 5 expands the categories of conflict of interest that paralegals need to understand.
- Chapter 6 adds material on the constitution (e.g., the Bill of Rights) as a foundation of our legal system.
- Chapter 7 has been reorganized to provide more comprehensive treatment of the basic components of legal analysis.
- Chapter 10 introduces predictive coding in the increasingly important arena of e-discovery of big data.

- Chapter 11 has been substantially reorganized in order to focus on the techniques of online legal research, particularly in the free databases.
- Chapter 12 confronts the dangers of copy-and-paste writing in an era when it is deceptively easy to rely on material that is available on the Internet.
- Chapter 12 also adds material on effective email writing.
- Chapter 13 catalogs the variety of social-media sites on the Internet and broadens the scope of security concerns that must be addressed.
- Chapter 13 introduces IoT: the Internet of things.
- Chapter 13 summarizes the major ethical issues covered in the book that result from computer use and misuse in the practice of law.
- The fifty-state survey of ethics in Appendix D expands the coverage of unauthorized practice of law (UPL).
- A new appendix (F) has been added on paralegal blogs.
- Appendix H covers major ways to locate different kinds of law offices as a guide in the job search.

#### **ANCILLARY MATERIALS**

#### **Instructor Companion Site** (



The online Instructor Companion Site provides the following resources:

#### Instructor's Manual

An Instructor's Manual and Test Bank by the author of the text accompany this edition and have been greatly expanded to incorporate changes in the text and to provide comprehensive teaching support. They include the following:

- Class ideas, such as lecture ideas and suggestions for using selected assignments.
- Al test bank of 645 questions, which includes a variety of questions in true/false, multiple-choice, and essay format. An answer key is also provided.

#### **PowerPoint Presentations**

Customizable Microsoft PowerPoint® Presentations focus on key points for each chapter. (Microsoft PowerPoint® is a registered trademark of the Microsoft Corporation.)

#### Cengage Learning Testing Powered by Cognero is a flexible, online system that allows you to:

- author, edit, and manage test bank content from multiple Cengage Learning solutions.
- create multiple test versions in an instant.
- deliver tests from your LMS, your classroom, or wherever you want.

#### Start right away!

Cengage Learning Testing Powered by Cognero works on any operating system or browser.

- No special installs or downloads needed
- Create tests from school, home, the coffee shop anywhere with Internet access

#### What will you find?

- Simplicity at every step. Al desktop-inspired interface features drop-down menus and familiar, intuitive tools that take you through content creation and management with ease.
- Full-featured test generator. Create ideal assessments with your choice of 15 question types (including true/false, multiple choice, opinion scale/Likert, and essay). Multi-language support, an equation editor and unlimited metadata help ensure your tests are complete and compliant.
- Cross-compatible capability. Import and export content into other systems.

To access additional course materials, please go to login.cengage .com, then use your SSO (single sign on) login to access the materials.



MindTap for Statsky/Introduction to Paralegalism is a highly personalized fully online learning platform of authoritative content, assignments, and services offering you a tailored presentation of course curriculum created by your instructor. MindTap for Statsky/Introduction to Paralegalism Law guides you through the course curriculum via an innovative learning path where you will complete reading assignments, annotate your readings, complete homework and engage with quizzes and assessments. MindTap includes a variety of web-apps known as "MindApps" - allowing functionality like having the text read aloud to you as well as MindApps that allow you to synchronize your notes with your personal Evernote account. MindApps are tightly woven into the MindTap platform and enhance your learning experience.

#### How MindTap helps students succeed

- Use the Progress App to see where you stand at all times—individually and compared to highest performers in your class.
- ReadSpeaker reads the course material to you.
- MyNotes provides the ability to highlight text and take notes - that link back to the MindTap material for easy reference when you are studying for an exam or working on a project.
- Merriam Webster Dictionary and a glossary are only a click away.

- Flashcards are pre-created to help you memorize the key terms.
- Drop boxes are provided for submitting instructorgraded exercises.

#### Not using MindTap in your course?

- It's an online destination housing ALL your course material and assignments ... neatly organized to match your syllabus.
- It's loaded with study tools that help you learn the material more easily.
- To learn more go to www.cengage.com/mindtap or ask your instructor to try it out.

#### **ACKNOWLEDGMENTS**

It is difficult to name all the individuals who have provided guidance in the preparation of the eight editions of this book. Looking back over the years, a number of people have played important roles in my initiation and growth as a student of paralegal education. I owe a debt to Jean and Edgar Cahn, founders of the Legal Technician Program at Antioch School of Law, where I worked; Bill Fry, Director of the National Paralegal Institute and a valued colleague since our days together at Columbia

Law School, where he was my dean in one of the first paralegal training programs in the country, the Program for Legal Service Assistants; Dan Oran, who helped me plan the first edition; Michael Manna, Ed Schwartz, Bill Mulkeen, Juanita Hill, Willie Nolden, and Linda Saunders.

I wish to thank the following people at Cengage Learning: Paul Lamond, Product Manager; Melissa Riveglia, Senior Content Developer; Diane Chrysler, Senior Product Assistant; Betty Dickson, Senior Content Project Manager; and Scott Chrysler, Marketing Manager.

Finally, a word of thanks to the reviewers who made valuable suggestions for improving the text:

Libby Pace California State University Los Angeles, CA

Olga A. Possé SUNY Buffalo / Millard Fillmore College Buffalo, NY

Casey D. Thompson Debbie Vinecour SUNY Rockland CC Suffern, NY [The] expanded use of well-trained..."paralegals," has been an important development. Today there are ... double the number of ... schools for training paralegals [as the number of schools for training attorneys].... The advent of the paralegal enables law offices to perform high quality legal services at a lower cost.

Possibly we have only scratched the surface of this development.

Warren E. Burger, Chief Justice of the United States Supreme Court, February 3, 1980

Paralegals are an absolutely essential component of quality legal services in the future.

\*James Fellers, President, American Bar Association, April 4, 1975

The court also commends the firm of Stull, Stull & Brody for the extensive use of paralegals rather than attorneys for various tasks, reducing the cost of litigation.

U.S. District Court Judge Helen Berregan, Feinberg v. Hibernia Corp., 966 F. Supp. 442, 448 (E.D. La., 1997)

Employment of paralegals and legal assistants is projected to grow 17 percent from 2012 to 2022, faster than the average for all occupations. Employers are trying to reduce costs and increase the availability and efficiency of legal services by hiring paralegals to perform tasks once done by lawyers. Paralegals are performing a wider variety of duties, making them more useful.

U.S. Department of Labor, Occupational Outlook Handbook (2010, 2014) (www.bls.gov/oob/legal/paralegals-and-legal-assistants.btm)

# HOW TO STUDY LAW IN THE CLASSROOM AND ON THE JOB

#### OUTLINE

- Classroom Learning
- On-the-Job Learning: The Art of Being Supervised

#### OBJECTIVES

After completing this section, you should be able to

- Know why legal education is a lifelong pursuit.
- Distinguish between substantive and procedural law, and rules and skills.
- Understand the importance of a study plan.
- Do a self-assessment of your study habits.
- Do a self-assessment of grammar, spelling, and composition skills.

- Create a self-improvement plan on the basics of writing.
- Know the importance of definitions in the law.
- Increase your note-taking skills.
- Maximize your learning opportunities on the job.
- Use a checklist for major assignments.

- Turn instructions into checklists.
- Know the value of models, background research, new publication lists, feedback, continuing legal education, and evaluations.

#### **CLASSROOM LEARNING**

Education does not come naturally to many of us. It is a struggle. This is all the more true for someone entering a totally new realm of training such as legal education. Much of the material may seem foreign and difficult. There is a danger of becoming overwhelmed by the vast quantity of laws and legal material that confront you. How do you study law? How do you learn law? What is the proper perspective that a student of law should have about the educational process? These are our concerns in this introduction to the process of studying law. In short, our theme is training to be trained—the art of effective learning.

The first step is to begin with a proper frame of mind. Too many students have false expectations of what legal education can accomplish. This can interfere with effective studying.

1. Your Legal Education Has Two Phases. Phase I Begins Now and Ends When You Complete This Training Program. Phase II Begins When This Training Program Ends and Is Not Completed Until the Last Day of Your Employment as a Paralegal.

You have entered a career that will require you to be a perpetual student. The learning never ends. This is true not only because the boundary lines of law are vast, but also because the law is changing every day. No one knows all of the law. Phase I of your legal education is designed to provide you with the foundation that will enable you to become a good student in phase II.

2. Your Legal Education Has Two Dimensions: The Content of the Law (the Rules) and the Practical Techniques of Using That Content in a Law Office (the Skills).

#### **Rules**

There are two basic kinds of rules or laws:

- Substantive Law1 The nonprocedural rules that define or govern rights and duties, e.g., the requirements for the sale of land and the elements of battery.
- Procedural Law: The rules that govern the mechanics of resolving a dispute in court or in an administrative agency, e.g., a rule on the time by which a party must respond to a complaint.

The law library contains millions of substantive and procedural laws written by courts (in volumes called *reporters*), by legislatures (in volumes called *statutory codes*), and by administrative agencies (in volumes called *administrative codes*). All great deal of the material in these volumes is also available on the Internet, as we will see. All substantial portion of your time in school will involve study of the substantive and procedural law of your state, and often of the federal government as well.

#### **Skills**

By far the most important dimension of your legal education will be the skills of using rules. Without the skills, the content of rules is close to worthless. Examples of legal skills include

- How to analyze the facts of a client's case in order to identify legal issues (see Chapter 7)
- How to interview a client (see Chapter 8)
- How to investigate the facts of a case (see Chapter 9)
- How to draft a complaint, the document that initiates a lawsuit (see Chapter 10)
- How to digest or summarize data found in discovery documents (see Chapter 10)
- How to do a cite check or perform other legal research in traditional and online law libraries (see Chapter 11)
- How to write a search query for an online database (see Chapter 13)

The overriding skill that, to one degree or another, is the basis for all others is the skill of legal analysis (Chapter 7). Some make the mistake of concluding that legal analysis is the exclusive domain of the attorney. Without an understanding of at least the fundamentals of legal analysis, however, paralegals cannot understand the legal system and cannot intelligently carry out many of the more demanding tasks they are assigned.

# 3. You Must Force Yourself to Suspend What You Already Know About the Law in Order to Be Able to Absorb (a) That Which Is New and (b) That Which Conflicts with Your Prior Knowledge and Experience.

Place yourself in the position of training students to drive a car. Your students undoubtedly already know something about driving. They have watched others drive and maybe have even had a lesson or two from friends. It would be ideal, however, if you could begin your instruction from point zero. There is a very real danger that the students have picked up bad habits from others. This may interfere with their capacity to listen to what you are saying. The danger is that they will block out anything you say that does not conform to previously learned habits and knowledge. If the habits or knowledge are defective, your job as a trainer is more difficult.

The same is true in studying law. Everyone knows something about the law from taking government or civics courses as a teenager and from the various treatments of the law in the media. Some of you may have been involved in the law as a party or as a witness in court. Others may have worked, or currently work, in law offices. Will this prior knowledge and experience be a help or a hindrance to you in your future legal education? For some, it will be a help. For many of us, however, there is a danger of interference.

This is particularly so with respect to the portrayal of the law on TVland in the movies. Contrary to what some TV programs may lead you to believe, it is highly unlikely that a judge would say to a witness, "I wouldn't believe you if your tongue came notarized." Furthermore, TV and movies sometimes give the impression that all attorneys are trial attorneys and that most cases are solved by dramatically tricking a hostile witness on the stand into finally telling the truth. Not so.

The practice of law is not an endless series of confessions and concessions that are pried loose from opponents. Every attorney does not spend all day engaged in the kind of case that makes front-page news. Recently a New Jersey paralegal left her job with a sole practitioner to take another paralegal position with a law firm that she thought was going to be like the one on the TV drama she faithfully watched every week. Three months later, she begged her old boss to take her back after discovering the huge gap between reality and the law office on that show.

Another potentially misleading portrayal of the law came in the O.J. Simpson criminal and civil trials, which captivated the nation in the late 1990s. Very few parties to litigation have teams of attorneys, investigators, and experts ready to do battle with each other as they did in the Simpson trials. The vast majority of legal disputes are never litigated in court. Most are either settled or simply dropped by one or both parties. Of the small number that are litigated, most involve no more than two opposing attorneys and several witnesses. In short, it is rare for the legal system to become the spectacle—some would say the circus—that the occasional high-profile case leads us to believe is common in the practice of law. While excitement and drama can be part of the legal system, they are not everyday occurrences. What is dominant is painstaking and meticulous hard work. This reality is almost never portrayed in the media.

Therefore, it is strongly recommended that you place yourself in the position of a stranger to the material you will be covering in your courses, regardless of your background and exposure to the field. Cautiously treat everything as a new experience. Temporarily suspend what you already know. Resist the urge to pat yourself on the back by saying, "I already knew that" or "I already know how to do that." For many students, such statements lead to relaxation. They do not work as hard once they have convinced themselves that there is nothing new to learn. No problem exists, of course, if these students are right. The danger, however, is that they are wrong or that they are only partially right. We are not always the best judges of what we know and of what we can do. Do not become too comfortable. Adopt the following healthy attitude: "I've heard about that before or I've already done that, but maybe I can learn something new about it." Every new instructor, every new supervisor, every new setting is an opportunity to add a dimension to your prior knowledge and experience. Be open to these opportunities. No two people practice law exactly the same way. Your own growth as a student and as a paralegal will depend in large part on your capacity to listen for, explore, and absorb this diversity.

#### 4. Be Sure That You Know the Goals and Context of Every Assignment.

Throughout your education, you will be given a variety of assignments: class exercises, text readings, drafting tasks, field projects, research assignments, etc. You should ask yourself the following questions about each one:

- What are the goals of this assignment? What am I supposed to learn from it?
- How does this assignment fit into what I have already learned? What is the context of the assignment?

For assignments from your textbook, carefully examine the table of contents, chapter objectives, and the variety of headings in a chapter in order to grasp the broader picture into which an assignment fits. Ask questions if the purpose of an assignment and its relationship to paralegalism is not clear. Successfully completing an assignment depends in part on understanding its goals and how these goals relate to the overall context of the course and the career.

#### 5. Design a Study Plan.

Make current lists of everything that you must do in each of your courses. Update the lists regularly. Divide every list into long-term projects (what is due next week or at the end of the semester) and short-term projects (what is due tomorrow). Have a plan for each day. Establish the following norm for yourself:

Every day you will make some progress on everything on your long-term and short-term lists.

Priority, of course, will be given to the short-term tasks. Yet some time, however small, will also be devoted to the long-term tasks. For example, on a day that you will be mainly working on the short-term projects, try to set aside 5 percent of your time for a long-term project by doing some background reading or by preparing a very rough first draft of an outline. Maybe all you can do on an assignment is read a paragraph or review what needs to be done on the assignment. It is critical that you establish momentum toward the accomplishment of all your tasks. This is done by never letting anything sit on the back burner. Set yourself the goal of making at least some progress on everything every day. Without this goal, momentum may be difficult to achieve and sustain.

Once you have decided what tasks you will cover on a given study day, the next question becomes the order in which will you cover them. There are a number of ways in which you can classify the things you must do. For example, you can classify them as

- easy tasks that will require a relatively short time to complete,
- complex tasks requiring more time, and
- tasks with time demands that will be unknown until you start them.

At the beginning of your study time, spend a few moments preparing an outline of the order in which you will cover the tasks that day and the approximate amount of time that you will set aside for each task. You may want to start with some of the easier tasks so that you can feel a sense of accomplishment relatively soon. Alternatively, you may want to devote early study time to the third kind of task listed above so that you can obtain a clearer idea of the scope and difficulty of such assignments. The important point is that you establish a schedule. It does not have to be written in stone. Quite the contrary. It is healthy to have enough flexibility to revise your day's schedule so that you can respond to unfolding realities as you study. Adaptation is not a sign of disorganization, but the total absence of an initial plan often is.

# 6. Add 50 Percent to the Time You Initially Think You Will Need to Study a Subject.

You are kidding yourself if you have not set aside a substantial amount of time to study law outside the classroom. The conscientious study of law takes time—lots of it. It is true that some students must work or take care of family responsibilities. You cannot devote time that you do not have. Yet the reality is that limited study time can lead to limited education.

Generally, people will find time for what they *want* to do. You may *wish* to do many things for which there will never be enough time. You will find the time, however, to do what you really want to do. Once you have decided that you want something badly enough, you will find the time to do it.

How much of each of your work hours is productive time? For most of us, the answer is about twenty minutes. The rest of the hour is spent worrying, relaxing, repeating ourselves, socializing, checking and posting to social media, etc. One answer to the problem of limited time availability is to increase the amount of productive time that you derive out of each work hour. You may not be able to add new hours to the clock, but you can add to your net productive time. How about moving up to thirty minutes an hour? Forty? You will be amazed at the time that you can "find" simply by making a conscious effort to remove some of the waste. When asked how a masterpiece was created, a great sculptor once responded: "You start with a block of marble and you cut away everything that is not art." In your study habits, start with a small block of time and work to cut away everything that is not productive.

In addition, look for ways to fit study time into your other activities. Always carry something to study in the event that time becomes available. For example, photocopy, scan, or photograph a portion of a chapter in a class textbook and bring it with you whenever you can. It might be perfect for that

unexpectedly long wait at the dentist's office. Constantly be on the alert for ways to increase the time you have available or, more accurately, to increase the productive time that you can make available.

There are no absolute rules on how much time you will need to study law. It depends on the complexity of the subject matter you must master. It is probably accurate to say that most of us need to study more than we do—as a rule of thumb, about 50 percent more.

Resolving time management problems as a student will be good practice for you when you are confronted with similar (and more severe) time management problems as a working paralegal. Many law offices operate at a hectic pace. One of the hallmarks of a professional is a pronounced reverence for deadlines and the clock.

Soon you will be gaining a reputation among other students, instructors, supervisors, and employers. You should make a concerted effort to acquire a reputation for hard work, punctuality, and conscientiousness about the time you devote to your work. In large measure, success follows from such a reputation. It is as important, and sometimes more important, than raw ability or intelligence. Phrased another way, your legal skills are unlikely to put bread on the table if you are casual about the clock.

#### 7. Create Your Own Study Area Free from Distractions.

It is essential that you find study areas that are quiet and free from distractions. Otherwise, concentration is obviously impossible. It may be that the worst places to study are at home or at the library unless you can find a corner that is cut off from noise and people who want to talk. If possible, study away from phones, TVs, iPods, and instant computer messaging. Do not make yourself available, except for emergencies. If you study in the corridor, at the first table at the entrance to the library, or at the kitchen table, you are inviting distraction. You need to be able to close yourself off for two to three hours at a time. It is important for you to interact with other people—but not while you are studying material that requires considerable concentration. You will be tempted to digress and to socialize. You are in the best position to know where these temptations are. You are also the most qualified person to know how to avoid the temptations.

Try this idea: never study with your smartphone within reach. Alternatively, shut the phone off when studying. If this is too radical, allow yourself no more than two minutes every hour to turn the phone on in order to find out if a true emergency demands your immediate attention.

## 8. Conduct a Self-Assessment of Your Prior Study Habits and Establish a Program to Reform the Weaknesses.

If you were to describe the way you study, would you be proud of the description? Here is a partial list of some of the main weaknesses of attitude or practice that students have about studying:

- They have done well in the past with only minimal study effort. Why change now?
- Others in the class do not appear to be studying very much. Why be different?
- They learn best by listening in class. Hence, instead of studying on their own, they wait until someone explains the material in person.
- They simply do not like to study; there are more important things to do in life.
- They can't concentrate.
- They study with lots of distractions, e.g., phones, radios, TVs, computers, iPods. Multitasking is more fun than unitasking.
- They get bored easily. "I can't stay motivated for long."
- They do not understand what they are supposed to study.
- They skim read.
- They do not stop to look up strange words or phrases.
- They study only at exam time—then cram for exams.
- They do not study at a consistent pace. They spend an hour (or less) here and there and have no organized, regular study times.
- They do not like to memorize.
- They do not take notes on what they are reading.

What other interferences with effective studying can you think of? Or, more important, which of the above items apply to you? How do you plead? In law, it is frequently said that you cannot solve a problem until you obtain the facts. What are the facts in the case of your study habits? Make your personal list of attitude problems, study patterns, and environmental interferences. Place these items in some order. Next, establish a plan for yourself. Which item on the list are you going to try to correct tonight? What will the plan be for this week? For next week? For the coming month? What specific steps will you take to try to change some bad habits? If one corrective method does not work, try

another. If the fifth does not work, try a sixth. Discuss techniques of improvement with other students and with instructors. Run this search in Google, Bing, or Yahoo: effective study habits. Be a sponge for self-improvement ideas. Prove to yourself that change is possible.

#### 9. Engage in Active Studying.

Even without distractions, the mind often wanders away from what should be the focus of what we are studying. Daydreaming during study time is not uncommon. If this occurs, do some active studying by forcing yourself to do something other than (or in addition to) reading. For example,

- Write out definitions of key terms.
- Create your own chart or graph that contains pieces or components of a topic you are studying.
- Create mnemonics to help you remember important concepts. (On mnemonics, see Section 16 below.)

# 10. Conduct a Self-Assessment on Grammar, Spelling, and Composition. Then Design a Program to Reform Weaknesses.

The legal profession lives by the written word. Talking is important for some activities, but writing is crucial in almost every area of law. You cannot function in this environment without a grasp of the basics of grammar, spelling, and composition. Al major complaint of employers today is that workers consistently violate these basics. The problem is serious. Here are seven steps that will help solve it:

#### **Step One**

Take responsibility for your training in grammar, spelling, and composition. Do not wait for someone to teach (or reteach) you the basics. Do not wait until someone points out your weaknesses. Make a personal commitment to train yourself. If English courses are available to you, great. Alweekly class, however, may not be enough.

#### **Step Two**

Raise your consciousness about the writing around you. When you are reading a newspaper, for example, try to be conscious of semicolons and paragraph structure in what you are reading. At least occasionally ask yourself why a certain punctuation mark was used by a writer. You are surrounded by writing. You read this writing for content. Begin a conscious effort to focus on the structure of the writing as well. This dual level of observation exists in other aspects of our lives. When people come out of the theater, for example, they often comment about how impressive the acting was. In addition to following the story or content of the play or movie, they were aware of its form and structure. These same levels of consciousness (content and form) should be developed for everything you read.

#### **Step Three**

Commit yourself to spending ten minutes a day, five or six days a week, on improving your grammar skills. For a total of about an hour a week, drill yourself on the fundamentals of our language. In Google, Bing, or Yahoo, run this search: *English grammar*. Do some surfing to find grammar sites that give basic rules and practice drills that allow you to test yourself on what you know. Avoid sites that charge fees or that are simply selling books. Here are some examples of sites you might find useful:

- www.edufind.com/english/grammar
- andromeda.rutgers.edu/~jlynch/Writing/c.html
- englishplus.com/grammar
- www.tnellen.com/cybereng/32.html

Sites such as these may provide links to comparable sites. Try several. Ask fellow students what sites they have found helpful.

How do you know what areas of grammar you should study? In Google, Bing, or Yahoo, run this search: *grammar self-test*. Find free tests. Take more than one. Make a list of areas where you need improvement. Start eliminating weaknesses one at a time.

One way to test your progress is to try the exercises on different sites. Suppose, for example, you are on a site that discusses the use of commas in *that* clauses and *which* clauses. After you finish the that/which exercises at this site, go to another grammar site and find its section on that/which. Read the examples and do the exercises at this site. Are you reinforcing what you learned at the first site, or are the examples and exercises at the second site making you realize that you need more work

understanding this area of grammar? Using more than one site in this way will help you assess how well you are grasping the material.

#### **Step Four**

Improve your spelling. Use a dictionary often. Begin making a list of words that you are spelling incorrectly. Work on these words. Ask other students, relatives, or friends to test you on them by reading the words to you one by one. Spell the words aloud or on paper. You can drill yourself into spelling perfection, or close to it, by this method. When you have the slightest doubt about the spelling of a word, check the dictionary. Add difficult words to your list. Again, the more often you take this approach now, the less often you will need to use the dictionary later.

Use the Internet as a resource. In Google, Bing, or Yahoo, run this search: spelling rules. You will be led to sites that will provide guidance.

Many word processors (e.g., Microsoft Word) have spell checkers that not only identify words you may have misspelled but also provide suggested corrections. Does this new technology mean that your spelling problems have been solved forever? Hardly. Spell checkers can catch many spelling blunders, but they can be very misleading. First of all, they cannot tell you how to spell many proper names, such as the surnames of individuals (unless you add these names to the base of words being checked). An even more serious problem is that spell checkers do not alert you to improper word choices. Every word in the following sentence, for example, is incorrect, but a spell checker would tell you that the sentence has no spelling problems:

"Its to later too by diner."

Here is what should have been written:

"It's too late to buy dinner."

Because the first sentence has no misspellings, you are led—that is, misled—to believe that you have written a flawless sentence.

#### **Step Five**

Enroll in English and writing courses. Find out if drop-in help labs or remedial centers are available to you. Check offerings at local schools, such as adult education programs in the public schools or at colleges. Call your local public library and ask what resources are available in the community.

#### **Step Six**

Find out which law courses in your curriculum require the most writing from students. If possible, take these courses, no matter how painful you find writing to be. In fact, the more painful it is, the more you need to place yourself in an environment where writing is demanded of you on a regular basis.

#### **Step Seven**

Simplify your writing. Cut down the length of your sentences. Often a long-winded sentence can be effectively rewritten into two shorter sentences. How can you tell if your sentences are too long? There are, of course, no absolute rules that will answer this question. Yet there is a general consensus that sentences on legal topics tend to be too long.

Several writing scholars have devised readability formulas that allow you to measure the readability of your writing in terms of how difficult it is to read. Among the most popular are the Gunning Fog Index, the Flesch Reading Ease Score, and the Flesch-Kincaid Grade Level Score. You can find them described on many Internet sites such as the following:

- en.wikipedia.org/wiki/Gunning-Fog\_Index
- en.wikipedia.org/wiki/Flesch-Kincaid\_Readability\_Test
- www.nightscribe.com/Education/eschew\_obfuscation.htm

Try a readability formula on your own writing. It will take only a few minutes to apply. Find out if the word processor you use (e.g., Word or WordPerfect) has a formula built into the program that is ready to use every time you write something.

Readability formulas are no more than rough guides. Use them to help raise your consciousness about your writing. As you review sentences you have written, you should be asking yourself self-editing questions such as "Would I have been clearer if I had made that sentence shorter?" The conscientious use of readability formulas will help you ask and answer such questions.

For an online manual on writing, including material on readability, see *A Plain English Handbook*, published by the Securities and Exchange Commission (www.sec.gov/pdf/handbook.pdf). Although this manual focuses on financial document filings, it contains many useful guidelines for any kind of writing.

To recap, prepare a self-assessment of weaknesses and set a schedule for improvement. Set aside a small amount of time each day to work on writing weaknesses. Be consistent about this time. Do not wait for the weekend or for next semester when you will have more time. The reality is that you will probably never have substantially more time than you have now. The problem is not so much the absence of time as it is an unwillingness to dig into the task. Progress will be slow and you will be on your own. Hence there is a danger that you will find "good" reasons (excuses) to do something else.

#### 11. Consider Forming a Student Study Group, but Be Cautious.

Students sometimes find it useful to form study groups. All healthy exchange with your colleagues can be very productive. One difficulty is finding students with whom you are compatible. Trial and error may be the only way to identify such students. All more serious concern is trying to define the purpose of the study group. It should not be used as a substitute for your own individual study. Course review, however, is an appropriate group task. Divide a course into parts, with each member of the group having responsibility for reviewing his or her assigned part with the rest of the group.

## 12. Use Your Legal Research Skills to Help You Understand Components of a Course That Are Giving You Difficulty.

The law library is more than the place to go to find law that governs the facts of a client's case. Algreat deal of the material in traditional and online law libraries consist of explanations, summaries, and overviews of the same law that you will be covering in your courses. See Exhibit 11-11 in Chapter 11 on doing relatively quick background research on any legal topic. Such research will be invaluable (especially the "dash searches" described in Exhibit 11-11) as outside reading to help resolve conceptual and practical difficulties you are having in class.

#### 13. Organize Your Learning Through Definitions or Definitional Questions.

Among the most sophisticated questions an attorney or paralegal can ask are these:

- What does that word or phrase mean?
- Should it be defined broadly or narrowly?

To a very large extent, the practice of law is a probing for definitions of key words or phrases in the context of facts that have arisen.

- Can a five-year-old be liable for negligence? (What is negligence?)
- Can the government tax a church-run bingo game? (What is the free exercise of religion?)
- Can attorneys in a law firm go on strike and obtain the protection of the National Labor Relations Act? (What is a covered employee under the labor statute?)
- Can one spouse rape another? (What is the definition of rape?)
- Can a citizen slander the president of the United States? (What is a defamatory statement?) Etc.

In every course that you take, you will come across numerous technical words and phrases in class and in your readings. Begin compiling a list of the major words and phrases for each class. When in doubt about whether to include something on your list, resolve the doubt by including it.

Then pursue definitions. Find definitions in class lectures, your textbook, a legal dictionary, or a legal encyclopedia. (For legal dictionaries on the Internet, type "legal dictionary" in Google, Bing, or Yahoo.) See also the online dictionary links in Exhibit 11-11 in Chapter 11.

For some words, you may have difficulty obtaining definitions. Do not give up your pursuit. Keep searching. Keep probing. Keep questioning. For some words, there may be more than one definition. Others may require definitions of the definitions.

Of course, you cannot master a course simply by knowing the definitions of all the key words and phrases covered in the course. Yet these words and phrases are the vocabulary of the course and are the foundation and point of reference for learning the other aspects of the course. Begin with vocabulary.

Consider starting a system of three-by-five or two-by-three cards or sheets of paper to help you learn the definitions. On one side, place a single word or phrase. On the other side, write the definition with a brief page reference or citation to the source of the definition. Using the cards or sheets, test yourself periodically. If you are in a study group, ask other members to test you. Ask a relative to test you. Establish a plan of ongoing review.

#### 14. Studying Ambiguity—Coping with Unanswered Questions.

Legal studies can be frustrating because there is so much uncertainty in the law. Legions of unanswered questions exist. Definitive answers to legal questions are not always easy to find, no matter how good your research techniques are. Every new fact situation presents the potential for a new law. Every law seems to have an exception. Furthermore, advocates frequently argue for exceptions to the exceptions. As indicated, when terms are defined, the definitions often need definitions.

The study of law is in large measure an examination of ambiguity that is identified, dissected, and argued. Search for as much clarity as you can, but do not be surprised if the conclusion of your search brings further questions. At time-honored answer to many legal questions is, "It depends!" Become familiar with the following frequently used equation:

If "fact X" is present, then the conclusion is "A," but if "fact Y is present," then the conclusion is "B," but if "fact Z" is....

The practice of law may sometimes appear to be an endless puzzle. Again, look for precision and clarity, but do not expect the puzzles to disappear.

#### 15. Develop the Skill of Note Taking.

Note taking is an important skill for getting the most out of your formal education. In addition, effective note taking is often a precondition to many law-office tasks.

Take notes on what you are reading for class preparation and for research assignments. Never rely exclusively on your memory. After reading hundreds of pages (or more) in textbooks and online, you will not be able to remember what you have read at the end of the semester, or even at the end of the day. Copy what you think are the essential portions of the materials you are reading. Include definitions of important words and phrases as indicated in Guideline 13 above.

In addition to (and not as a substitute for) taking written notes on your readings, mark up the text if you own it. Use felt pens of different colors. Underline and highlight what appears to be important. Circle key words. Don't read passively. Read with your eyes and hands!

To be sure, note taking will add time to your studying. Yet you will discover that it was time well spent, particularly when you begin reviewing for an exam or writing a memorandum.

In class, you must develop the art of taking notes while simultaneously listening to what is being said. On the job, you will have to do this frequently, such as when

- Receiving instructions from a supervisor
- Talking with someone on the phone
- Doing a follow-up interview of a client
- Interviewing a witness during field investigation

If your supervisor asks you to attend a deposition or a trial, you may need to take detailed notes on the testimony of a particular witness. (A deposition, as you will learn in Chapter 10, is a method of discovery by which parties and their prospective witnesses are questioned outside the courtroom before trial.) Al good place to begin learning how to write and listen at the same time is during your classes.

Most students take poor class notes. This is due to a number of reasons:

- They write slowly.
- They don't like to take notes; it's hard work.
- They don't know if what is being said is important enough to be noted until after it is said—when it is too late because the instructor has gone on to something else.
- They don't think it's necessary to take notes on a discussion that the instructor is having with another student.
- They don't think it's necessary to take notes on a topic that's covered in the textbook.
- They take notes only when they see other students taking notes.
- Some instructors ramble.

Alstudent who uses these reasons for not taking comprehensive notes in class will eventually be using similar excuses on the job when precise note taking is required for a case. This is unfortunate. You must overcome whatever resistance you have acquired to the admittedly difficult task of note taking. Otherwise, you will pay the price in your schoolwork and on the job.

Develop your own shorthand system of abbreviations for note taking. Here are some commonly used abbreviations in the law. (Some of the terms have more than one abbreviation; pick the one that works best for you.)

aa         administrative agency         dba         doing business as fifted appellate court         juv         juvenile         rsb         reasonable and the part of t	a	action	d	danger/dangerous	jur	jurisdiction	rr	railroad
a/c         appellate court         d/e         direct examination         K         contract         s         sum           aff         affirmed         dept         department         I         liable, liability         S         statute           aka         also known as         df         defendant         II         landlord         S         suppose           ans         answer         dist         district         lit         litigation         sn/b         should not be           app         appellant         dob         date of birth         mfr         maximum         sn/b         should not be           appelle         dob         date of birth         mfr         manufacturing         s/f         statute of frauds           ar         administrative regulation         dos         date of separation         mfr         manufacturing         s/f         statute of frauds           ar         administrative regulation         dos         date of separation         mfr         manufacturing         s/f         statute of frauds           ar         assumption of risk         ce         employee         mkt         market         sl         strict lability           assumption of risk         c	aa	administrative agency	dba	doing business as	juv	juvenile	rsb	reasonable
aka         also known as         df         defendant         II         landlord         \$         suppose           ans         answer         dist         district         lit         litigation         s/b         should be           app         appeal         dmg         damages         max         maximum         sn/b         should not be           appen         appellant         dob         date of birth         mfr         manufacturer         sc         supreme court           appen         appellee         dod         date of separation         min         manufacturing         s/f         statute of frauds           ar         administrative regulation         dos         date of separation         min         min minimum         s/f         summary judgment           a/r         assumption of risk         ee         employee         mkt         market         sl         strict liability           ass         association         eg         example         >         more than; greater than         s/l         statute of limitations           atty         attrict liability         ass         statute of limitations           atty         attrict liability         statute         statute	a/c		d/e	direct examination	K	contract	s	sum
aka       also known as       df       defendant       II       landlord       \$       suppose         ans       answer       dist       district       lit       litigation       \$fb       should be         app       appeal       dmg       damages       max       maximum       snb       should not be         appen       appellen       dob       date of birth       mfr       manufacturer       sc       supreme court         appen       appellen       dod       date of birth       mfr       manufacturer       sc       supreme court         ar       administrative regulation       dos       date of separation       min       minimimm       s/f       statute of frauds         ar       administrative regulation       dos       date of separation       min       minimimm       s/f       statute of frauds         ar       assumption of risk       ee       employee       mkt       market       s1       strict liability         assn       association       eg       example       >       more than; greater than       s1       statute of finuitations         stty       attorney       egs       example       >       more than; greater than       s1	aff	affirmed	dept	department	1	liable, liability	S	statute
app app app app app appell appelle appelle appelle appelle appelle adate of birth dod date of death are administrative regulation a/r assumption of risk ee employee manufacturing dod date of death mfg manufacturing min minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum minimum mi	aka	also known as		defendant	11	landlord	\$	suppose
appet appellant appellant appellant appelle dod date of birth mfr manufacturier sc supreme court appee appelle dod date of death mfg manufacturing s/f statute of frauds ar administrative regulation a/r assumption of risk ee employee mkt market sl strict liability assn association eg example > more than; greater than s/l statute of limitations atty attorney egs examples > more than; greater than s/l statute of limitations atty attorney egs examples   equity   mun municipal   std   standard   b/c   because   equity   mun municipal   std   standard   sub-part   sub-par	ans	answer	dist	district	lit	litigation	s/b	should be
appee ar appellee ar abpellee appellee appellee abpellee ar abpellee abpellee administrative regulation a/r assumption of risk assn asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson asson <b< td=""><td>app</td><td>appeal</td><th>dmg</th><td>damages</td><th>max</th><td>maximum</td><th>sn/b</th><td>should not be</td></b<>	app	appeal	dmg	damages	max	maximum	sn/b	should not be
ar administrative regulation dos date of separation min minimum s/j summary judgment a/r assumption of risk ee employee mkt market sl strict liability assn association eg example > more than; greater than s/l statute of limitations atty attorney egs examples < less than; smaller than ss statute of limitations atty attorney eqs equity mun municipal std standard b/c because eqbl equitable n/a not applicable sub substantial b/k brach of contract er employer N negligence subj subject b/p burden of proof est estimate natl national t tort bfp bona fide purchaser ev evidence negl negligence t/c trial court ecomplaint 4cb foreseable # number tp third party ca court of appeals fed federal O owner vs against; versus c/a cause of action fs federal O owner vs against; versus ecomplaint gyt government p/c prosimate cause w/ with cc child custody h h holder in due course pg page w/i within cert certiorari indl individual pl plaintiff w/o without c/l common law indp independent pv privilege x cross-examination cr criminal ins insurance re real estate ? question cr criminal court intl international reg regulation + plus plaintiff court international cr civil jj judges, justices revd reversed ¶ plaintiff	appnt	appellant	dob	date of birth	mfr	manufacturer	sc	supreme court
a/rassumption of riskeeemployeemktmarketslstrict liabilityassnassociationegexample>more than; greater thanslstatute of limitationsattyattorneyegsexamples<less than; smaller thanssstatute of limitationsbb usinesseqequitymunmunicipalstdstandardb/cbecauseeqblequitablen/anot applicablesubsubstantialb/kbreach of contracteremployerNnegligencesubjsubjectb/pburden of proofestestimatenatlnationalttortbfpbona fide purchaserevevidenceneglnegligencet/ctrial court©considerationffactnjinjuryteetrusteeecomplaint4cbforeseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/wiftc/ccounterclaimgvtgovernmentp/cproximate causew/wiftc/ccross-examinationhusbandp/fprima faciewdwrongful deathc/ccross-examinat	appee	appellee	dod	date of death	mfg	manufacturing	s/f	statute of frauds
a/rassumption of riskeeemployeemktmarketslstrict liabilityassassociationegexample>more than; greater thans/lstatute of limitationsattyattorneyegsexamples<less than; smaller thanssstatute of limitationsbb usinesseqequitymunmunicipalstdstandardb/cbecauseeqblequitablen/anot applicablesubsubstantialb/kbreach of contracteremployerNnegligencesub;subjectb/pburden of proofestestimatenatlnationalttortbfpbona fide purchaserevevidencenegligencet/ctrial court©considerationffactnjinjuryteetrusteeccomplaint4cbfor eseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withccchild custodyhhusbandp/fprima faciew/w/withincecross-examinationhdcholder in due coursepgpagew/w/withincertce	ar	administrative regulation	dos	date of separation	min	minimum	s/j	summary judgment
attyattorneyegsexamples<less than; smaller thanssstate statutebbusinesseqequitymunmunicipalstdstandardb/cbecauseeqblequitablen/anot applicablesubsubstantialb/kbreach of contracteremployerNnegligencesub;subjectb/pburden of proofestestimatenatlnationalttortbfpbona fide purchaserevevidenceneglnegligencet/ctrial court©considerationffactnjinjuryteetrusteeccomplaint4cbforeseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withc/ccounterclaimgvtgovernmentp/fprima faciewdwrongful deathc/ccounterclaimgvtgovernmentp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithoutcertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindp <t< td=""><td>a/r</td><td>assumption of risk</td><th>ee</th><td></td><th>mkt</th><td>market</td><th>sl</th><td>strict liability</td></t<>	a/r	assumption of risk	ee		mkt	market	sl	strict liability
b         business         eq         equity         mun         municipal         std         standard           b/c         because         eqbl         equitable         n/a         not applicable         sub         substantial           b/k         breach of contract         er         employer         N         negligence         subj         subject           b/p         burden of proof         est         estimate         natl         national         t         tort           bfp         bona fide purchaser         ev         evidence         negligence         t/c         trial court           ©         consideration         f         fact         nj         injury         tee         trustee           c         complaint         4cb         foreseeable         #         number         tp         third party           ca         coupt of appeals         fed         federal         O         owner         vs         against; versus           c/a         cause of action         fs         federal statute         op         opinion         w         wife           c/c         counterclaim         gvt         government         p/c         proximate cause	assn	association	eg	example	>	more than; greater than	s/l	statute of limitations
b/c because eqbl equitable n/a not applicable sub substantial b/k breach of contract er employer N negligence subj subject b/p burden of proof est estimate natl national t tort bfp bona fide purchaser ev evidence negl negligence t/c trial court consideration f fact nj injury tee trustee complaint 4cb foreseeable # number tp third party ca court of appeals fed federal O owner vs against, versus c/a cause of action fs federal tatute op opinion w wife combined court child custody h h husband p/f prima facie wd wrongful death cert certiorari indl individual pl plaintiff w/o without coll common law indp independent pv privilege x cross examination cr criminal ins insurance re real estate ? question cr criminal court intl international reg regulation reversed ¶ plaintiff versersed full defendant court of plaintiff plaintiff plaintiff reversed full defendant plaintiff reversed full defendant plaintiff reversed full defendant representative full plaintiff reversed full plaintiff reversed full plaintiff representative full defendant representative full plaintiff representative full plaintiff representative full defendant representative full plaintiff representative full plaintiff representative full defendant representative full plaintiff representative full plaintiff representative full plaintiff representative full plaintiff representative full defendant representative full plaintiff representative full plaintiff representative full plaintiff representative full representative full plaintiff representative full plaint	atty	attorney	egs	examples	<	less than; smaller than	SS	state statute
b/kbreach of contracteremployerNnegligencesubjsubjectb/pburden of proofestestimatenatlnationalttortbfpbona fide purchaserevevidenceneglnegligencet/ctrial court©considerationffactnjinjuryteetrustee¢complaint4cbforeseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withc/ccounterclaimgvtgovernmentp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcross-examinationcrcriminalinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminalintentionalregregulation+pluscschild supportITintentionalrev <td>b</td> <td>business</td> <th>eq</th> <td>equity</td> <th>mun</th> <td>municipal</td> <th>std</th> <td>standard</td>	b	business	eq	equity	mun	municipal	std	standard
b/p bfpburden of proof bona fide purchaserest ev evidencenatl negl negligencenational negligencet tvc trial court© considerationf f factnj foreseeableinjurytee trustee¢ complaint4cb foreseeablefed federalO ownerownervs against; versusc/a cause of actionfs federal statuteop opinionw wifec/c counterclaimgvt governmentp/c proximate causew/ withc/c cohild custodyh h h holder in due course cert cert certiorari considerational law con l cross-examinationhdc holder in due course pg pagepage page page w/i withinc/l common law cross-indicational law cross-indicational law cross-indicational law cross-indicational law cross-criminal cross-criminal cross-criminal cross-criminal ins insurance insurancep resonable regulationx/e resonable regulationcross-child support ct court cv civilIT intentional tort intentional tort rep representative\ rev reverse\ rev reversed rev reverse\ rev reversed rev reverse\ rev reverse	b/c	because	eqbl	equitable	n/a	not applicable	sub	substantial
bfpbona fide purchaserevevidenceneglnegligencet/ctrial court©considerationffactnjinjuryteetrustee¢complaint4cbforeseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withc/ecointerclaimgvtgovernmentp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithinc/ecross-examinationhdcholder in due coursepgpagew/iwithoutc/lcommon lawindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcross-examinationcon1constitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminalintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourt <td< td=""><td>b/k</td><td>breach of contract</td><th>er</th><td>employer</td><th><math>\mathbf{N}</math></th><td>negligence</td><th>subj</th><td>subject</td></td<>	b/k	breach of contract	er	employer	$\mathbf{N}$	negligence	subj	subject
©considerationffactnjinjuryteetrustee¢complaint4cbforeseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withccchild custodyhhusbandp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcross-examinationc/lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverse\defendantcvciviljjjudges, justicesrevdreversed¶	b/p	burden of proof	est	estimate	natl	national	t	tort
¢complaint4cbforeseeable#numbertpthird partycacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withc/cchild custodyhhusbandp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcrosscon lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreversedAdefendantcvciviljjjudges, justicesrevdreversed¶plaintiff	bfp	bona fide purchaser	ev	evidence	negl	negligence	t/c	trial court
cacourt of appealsfedfederalOownervsagainst; versusc/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withccchild custodyhhusbandp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcrosscon lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverseΔdefendantcvciviljjjudges, justicesrevdreversed¶plaintiff	©	consideration	f	fact	nj	injury	tee	trustee
c/acause of actionfsfederal statuteopopinionwwifec/ccounterclaimgvtgovernmentp/cproximate causew/withccchild custodyhhusbandp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcrosscon lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverseΔdefendantcvciviljjjudges, justicesrevdreversed¶plaintiff	¢	complaint	4cb	foreseeable	#	number	tp	third party
c/ccounterclaimgvtgovernmentp/cproximate causew/withccchild custodyhhusbandp/fprima faciewdwrongful deathc/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcrosscon lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrcriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverse\defendantcvciviljjjudges, justicesrevdreversed¶plaintiff	ca	court of appeals	fed	federal	O	owner	vs	against; versus
cc       child custody       h       husband       p/f       prima facie       wd       wrongful death         c/e       cross-examination       hdc       holder in due course       pg       page       w/i       within         cert       certiorari       indl       individual       pl       plaintiff       w/o       without         c/l       common law       indp       independent       pv       privilege       x       cross         con l       constitutional law       info       information       ®       reasonable       x/e       cross-examination         cr       criminal       ins       insurance       re       real estate       ?       question         cr       criminal court       intl       international       reg       regulation       +       plus         cs       child support       IT       intentional tort       rep       representative       \lambda       defendant         ct       court       j       judges, justices       rev       reversed       ¶       defendant         cv       civil       jj       judges, justices       revd       reversed       ¶	c/a	cause of action	fs	federal statute	op	opinion	$\mathbf{w}$	wife
c/ecross-examinationhdcholder in due coursepgpagew/iwithincertcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcrosscon lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrccriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverse $\Delta$ defendantcvciviljjjudges, justicesrevdreversed $\P$ plaintiff	c/c	counterclaim	gvt	government		proximate cause	w/	
certcertiorariindlindividualplplaintiffw/owithoutc/lcommon lawindpindependentpvprivilegexcrosscon lconstitutional lawinfoinformation®reasonablex/ecross-examinationcrcriminalinsinsurancerereal estate?questioncrccriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverse $\Delta$ defendantcvciviljjjudges, justicesrevdreversed $\P$ plaintiff	cc	child custody	h	husband	p/f	prima facie	wd	
$c/l$ common lawindpindependent $pv$ privilege $x$ cross $con l$ constitutional lawinfoinformation $\mathfrak{B}$ reasonable $x/e$ cross-examination $cr$ criminalinsinsurance $re$ real estate?question $crc$ criminal courtintlinternational $reg$ regulation $+$ plus $cs$ child supportITintentional tort $rep$ representative $\lambda$ therefore $ct$ court $j$ judge, justice $rev$ reverse $\Delta$ defendant $cv$ civil $jj$ judges, justices $revd$ reversed $\P$ plaintiff	c/e	cross-examination	hdc	holder in due course	$\mathbf{pg}$	page	w/i	within
con l       constitutional law       info       info mation       ®       reasonable       x/e       cross-examination         cr       criminal       ins       insurance       re       real estate       ?       question         cr       criminal court       intl       international       reg       regulation       +       plus         cs       child support       IT       intentional tort       rep       representative       \( \)       therefore         ct       court       j       judge, justice       rev       reverse       \( \)       defendant         cv       civil       jj       judges, justices       revd       reversed       \( \)       plaintiff	cert	certiorari	indl	individual	pl	plaintiff	w/o	without
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	c/l	common law	indp	independent	pv	privilege	X	cross
crccriminal courtintlinternationalregregulation+pluscschild supportITintentional tortreprepresentative\thereforectcourtjjudge, justicerevreverse $\Delta$ defendantcvciviljjjudges, justicesrevdreversed $\P$ plaintiff	con l	constitutional law	info	information	®	reasonable	x/e	cross-examination
cschild supportITintentional tortreprepresentative\ thereforectcourtjjudge, justicerevreverse $\Delta$ defendantcvciviljjjudges, justicesrevdreversed $\P$ plaintiff	cr	criminal	ins	insurance	re	real estate	?	question
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	crc	criminal court	intl	international	reg		+	plus
cv civil jj judges, justices revd reversed ¶ plaintiff	cs	child support	IT	intentional tort	rep	representative	\	therefore
	ct	court		judge, justice	rev	reverse	Δ	defendant
	cv	civil	jj	judges, justices	revd	reversed	P	plaintiff
	cz	cause	jt	judgment				

If you can't hear the instructor, move closer or ask the instructor to speak up. Ask the instructor to repeat something you missed. Consider taping the class so that you can continue note taking while listening to the tape.

If you are participating in class by talking with the instructor, it will obviously be difficult for you to take notes at the same time. After class, take a few moments to jot down some notes on what occurred during the discussion. Then ask someone else who was in class to review these notes for accuracy and completeness.

Sometimes you will have to begin taking notes at the moment the person starts talking rather than wait until the end of what he or she is saying. Try different approaches to increasing the completeness of your notes.

Don't fill up entire sheets of paper with notes. Keep at least a three-inch right-hand margin that is blank while you take notes. Later, fill in this empty space with notes on things you missed in the lecture or that you better understood as the class progressed. Read the notes of fellow students. When their notes contain material you missed, copy it into the right-margin space.

Sometimes so much is happening and being said during a class that it is difficult to take notes. Things can sometimes get a bit chaotic. When this happens, at least write down what appear to be key nouns or verbs that the instructor uses. Leave additional space around these words so that you can come back later and fill in details and context that you may have missed initially.

Pay particular attention to definitions and lists. Every time the instructor defines something, be sure your notes record the definition exactly. Sometimes instructors will redefine the same term with slight modifications. Take notes on the modifications because they will often help clarify what is being discussed. Also, be conscientious about lists. An instructor might begin a topic by saying that "there are three elements" to a particular rule, or that "we will be examining four major categories of examples of the kinds of cases that can arise." When you hear that a list is coming by such language, have your pen at the ready. Lists of this kind are very important in the law.

#### 16. Studying Rules: The Role of Memory.

Memory plays a significant role in the law. Applicants for the bar, for example, are not allowed to take notes into the exam room. An advocate in court or at an administrative hearing may be able to refer to notes, but the notes are of little value if the advocate does not have a solid grasp of the case. Most of the courses you will be taking have a memory component. This is true even for open-book exams. You will not have time to go through all the course material while responding to the questions.

Students often make two mistakes on the role of memory:

- They think that memorizing is beneath their dignity.
- They think that because they understand something, they know it sufficiently to be able to give it back in an examination.

Of course, you should not be memorizing what you do not understand. Rote memorization is close to worthless. This is not so for important material that you comprehend. Yet simply understanding something does not necessarily mean that you have a sufficient grasp of it for later use.

Many systems for memorizing material can be effective:

- Reading it over and over
- · Copying and recopying important parts of it
- · Having other students ask you questions about it
- · Making summaries or outlines of it
- Tape-recording yourself reading difficult material from a textbook and playing the tape back while driving or doing house chores
- Creating your own mnemonics

Mnemonics are simply aids to memory that you create or adopt. The most common mnemonics consist of a series of letters that represent items on a list. For example, suppose you are studying section § 100, an important larceny statute in criminal law, which requires proof of the following three elements:

- Intent to steal
- Personal property
- · Resulting financial harm

To help you remember these elements, you could assign the letter "I" to the first element, "P" to the second, and "R" to the third. You can then scramble these letters to make the word RIP:

Resulting financial harm [R]

Intent to steal [I]

Personal property [P]

The word RIP can now be used to help you remember the elements of § 100. Easy-to-remember words (e.g., RIP) or nonwords (e.g., GGET) or ANAR) can be useful mnemonics. You are in control of the letters to be assigned to items on the list. Your goal is to come up with something that will help you recall the list; therefore, only criterion is to come up with something memorable!

If you do not have a photographic mind, you must resort to such techniques. Try different systems. Ask other students for tips on how they memorize material. For more ideas, run this search in Google, Bing, or Yahoo: *student improve memory*.

You will have to find out from your instructor what material you will be expected to know for the course. You can also check with other students who have had this instructor in the past. It may not always be easy to find out how much an instructor expects you to know from memory. Instructors have been known to surprise students on examinations!

#### 17. Studying Skills—the Necessity of Feedback.

Memory is most important when you are studying the basic principles of substantive and procedural law. Memory plays a less significant role in learning the skills of interviewing, investigation, legal analysis, drafting, coordinating, digesting, and advocacy. These skills have their own vocabulary that you must know, but it is your judgmental rather than your memory faculties that are key to becoming competent in such skills.

They are developed primarily by practice drills or exercises. The learning comes from the feedback that you obtain while engaged in the skill exercises. What are the ways to obtain feedback?

- Evaluations on assignments and exams
- Role-playing exercises that are critiqued in class
- Comparisons between your work (particularly writing projects) and models that are provided by the instructor or that you find on your own in the library
- Critiques that you receive from students in study groups

Be constantly looking for feedback. Do not wait to be called on. Do not wait to see what feedback is planned for you at the end of the course. Take the initiative.

- Seek conferences or email contact with your instructors.
- Find out who is available to read your writing or to observe your performance in any of the other skills.
- Set up your own role-playing sessions with your fellow students.
- Seek critiques of your rewrites even if rewriting was not required.
- Look for opportunities to critique other students on the various skills.
- Ask other students to let you read their graded examinations so that you can compare their papers with your own.
- Create your own hypotheticals for analysis in study groups. (Al hypothetical is simply a set of facts invented for the purpose of discussion and analysis.)
- Do additional reading on the skills.

In short, become actively involved in your own skill development.

#### 18. The Value of Speed-Reading Courses in the Study of Law.

In the study of law, a great deal of reading is required. Should you, therefore, take a speed-reading course? No, unless the course helps you slow down the speed of your reading! This advice may be quite distasteful to advocates (and salespersons) of speed-reading courses. The reality, however, is that statutes, regulations, and court opinions cannot be speed-read. They must be carefully picked apart and read word for word, almost as if you were translating from one language into another.

If you are troubled by how long it takes you to read, do not despair. Keep reading. Keep rereading. The pace of your reading will pick up as you gain experience. Never strive, however, to be able to fly through the material. Strive for comprehensiveness. Strive for understanding. For most of us, this will come through the slow process of note taking and rereading. It is sometimes argued that comprehension is increased through speed. Be wary of this argument. Reading legal material calls for careful thinking about what you read—and taking notes on these thoughts. There may be no harm in rapidly reading something for the first time. At your second, third, and fourth reading, however, speed is rarely helpful.

#### ON-THE-JOB LEARNING: THE ART OF BEING SUPERVISED

Algreat deal of learning will occur when you are on the job. Some of it may come through formal in-house office training and by the study of office procedure manuals. Most of the learning, however, will come in day-to-day interaction with your supervisors as you are given assignments. The learning comes through being supervised. Here are some guidelines to assist you in this important dimension of your legal education.

#### 1. Don't Play "Emperor's Clothes" with the Instructions That You Receive.

Recall the story of the emperor's clothes. He walked around without clothes, but everybody kept saying what a beautiful wardrobe he had. As new people arrived, they saw that he had no clothes, but they heard everyone talking as if he were fully dressed. The new people did not want to appear unintelligent, so they, too, began admiring the emperor's wardrobe. When paralegals are receiving instructions on an assignment, they play "emperor's clothes" when they pretend that they understand all the instructions but in fact do not. They do not want to appear to be uninformed. They do not want to give the impression that they are unsure of themselves. For obvious reasons, this is a serious mistake.

Whenever you are given an assignment in a new area—that is, an assignment on something that you have not done before—there should be a great deal that you do not understand. This is particularly true during your first few months on the job, when just about everything is new! Do not pretend to be something you are not. Constantly ask questions about new things. Do not be reluctant to ask for explanations. Do not assume that the instructions will become clear to you after you start the assignment. Ask for help at the outset. It will not be a sign of weakness. Quite the contrary. People

who take steps to make sure that they fully understand all the instructions they receive will soon gain a reputation for responsibility and conscientiousness.

#### 2. Repeat the Instructions to Your Supervisor Before You Leave the Room.

Once your supervisor has told you what he or she wants you to do, do not leave the room in silence or with the general observation "I'll get on that right away." Repeat the instructions back to the supervisor as you understand them. Make sure that you and your supervisor are on the same wavelength by summarizing what you think you were told to do. This will be an excellent opportunity for the supervisor to determine what you did or did not understand and to provide clarifications where needed.

Supervisors will not always be sure of what they want. By trying to obtain clarity on the instructions, you are providing them with the opportunity to think through what they want done. In the middle of the session with you, the supervisor may change his or her mind on what is to be done.

#### 3. Write Down Your Instructions.

Paralegal Jamie Collins has frequently supervised and trained other paralegals and clerical staff. "I cannot tell you," says Jamie, "how many people I have trained that would come back to ask me a question, and never write down a thing, only to return to my office a few moments or hours later or the next day with the exact same question, and no clue how to answer it."<sup>2</sup>

Never go to your supervisor without pen and paper. Preferably, keep an instructions notebook, diary, or journal in which you record the following information:

- Notes on what you are asked to do
- Whether the tasks in the assignment are billable (i.e., whether a particular client will be asked to pay fees for the performance of those tasks)
- The date you received the assignment
- The date by which the supervisor expects you to complete all or part of the assignment; if an exact due date is not provided, an estimate of the amount of time the task should take (sometimes referred to as a *time budget*)
- The date you actually complete the assignment
- Comments made by supervisors or others on what you submit (if no one makes comments, take the initiative and ask for feedback)

The notes will serve as your memory bank. Whenever any questions arise about what you were supposed to do, you have something concrete to refer to.

Exhibit Al contains an assignment checklist on which you can record this kind of data for every major assignment you receive.

#### **Exhibit A**

#### **Checklist for Major Assignments**

- · Name of supervisor for the assignment:
- What you have been asked to do:
- Identification of the client or matter for which the task is being done:
- Specific areas or tasks you have been told not to cover, if any:
- Format supervisor expects, e.g., computer text, outline only, rough draft, final copy ready for supervisor's signature:
- Date you are given the assignment:
- Expected due date:
- Time budget (the amount of time the task should take):

#### **Exhibit A**

#### Checklist for Major Assignments (Continued)

- Is the task billable to a client? If so, to what account? Is there a limit on the number of hours to spend on the assignment?
- Location of samples or models in the office to check as possible guides:
- Possible resource people in the office you may want to contact for help:
- Practice guide, formbook, or manual in the library that might provide background or general guidance:
- Dates you contacted supervisor or others for help before due date:
- Date you completed the assignment:
- Positive, negative, or neutral comments from supervisor or others on the quality of your work on the assignment:
- Things you would do differently the next time you do an assignment of this kind:

#### 4. Ask for a Due Date and a Statement of Priorities.

You need to know when an assignment is due. What's the time budget? Ask for a due date even if the supervisor tells you to "get to it when you can." This phrase may mean "relatively soon" or "before the end of the month" to your supervisor, but not to you. If the supervisor says he or she does not know when it should be done, ask for an approximate due date. Tell the supervisor you want to place the assignment on your calendar so that it can be completed in a timely manner along with all your other assignments. Ask what priority the assignment has. Where does it fit in with your other assignments? If you have more than enough to fill the day, you need to know what takes priority. If you do not ask for a priority listing, the supervisor may assume you are under no time pressures. If you have assignments from different supervisors, gently suggest that the supervisors confer and let you know what the priorities are.

# 5. If the Instructions Appear to Be Complicated, Ask Your Supervisor to Identify the Sequence of Tasks Involved.

As you receive instructions, you may sometimes feel overwhelmed by all that is being asked of you. Many supervisors do not give instructions in clear, logical patterns. They may talk in a rambling, stream-of-consciousness fashion. When confronted with this situation, simply say:

OK, but can you break that down for me a little more so that I know what you want me to do first? It would help if I approach it one step at a time. Where do you want me to start?

# 6. As Often as Possible, Write Your Instructions and What You Do in the Form of Checklists.

Al methodical mind is one that views a project in "doable" steps and that tackles one step at a time. You need to have a methodical mind in order to function in a busy law office. One of the best ways to develop such a mind is to think in terms of checklists. Attorneys love checklists. A great deal of the practice material published by bar associations, for example, consists of page after page of detailed checklists of things to do and/or to consider when completing a project for a client. Attorneys want to be thorough. An unwritten "rule" of law practice seems to be that you cannot be thorough without a checklist.

Alchecklist is simply a chronological sequencing of tasks that must be done in order to complete an assignment. Convert the instructions from your supervisor into checklists. In the process of actually carrying out instructions, you go through many steps—all of which could become part of a detailed checklist. The steps you went through to complete the task become a checklist of things to do in order to complete such a task in the future. To be sure, it can be time-consuming to draft checklists. Keep in mind, however, that

- The checklists will be a benefit to you in organizing your own time and in assuring completeness.
- The checklists can be invaluable for other employees who are given similar assignments in the future.
- Your supervisors will probably be very impressed by your initiative and organizational ability.

You will not be able to draft checklists for everything that you do. Perhaps you will not be able to write more than one checklist a week. Perhaps you will have to use some of your own time to write checklists. Whatever time you devote will be profitably spent so long as you are serious about writing and using the checklists. They may have to be rewritten or modified later. This should not deter you from the task. Most things that are worth doing require testing and reassessment.

Here is how one veteran paralegal describes the process:

When you are doing a multistep task (thinking out each step as you move through it)...create a checklist that will help you do it faster next time. [Also] create a notebook to store it in so that when the job comes around again you can find it! This is trickier to do on a busy day than you may think, but well worth doing. Think of these checklists as expanding your professional options. In time, they may become resources for training you give to newer paralegals.<sup>3</sup>

Many how-to manuals found in law offices were created out of the checklists that workers compiled on tasks they frequently performed.

#### 7. Find Out What Practice Guides and Checklists Already Exist in Your Office.

If practice guides and checklists on the topic of your assignment already exist in your office, you should find and use them. (Also check in computer databases where the office may store frequently used forms and instructions.) Unfortunately, the how-to-do-it information may be buried in the heads of the attorneys, paralegals, and secretaries in the office. No one may have taken the time to write it all down. If this is not so, find out where it is written down and try to adapt what you find to the assignment on which you are working.

#### 8. Ask for a Model.

One of the best ways to make sure you know what a supervisor wants is to ask whether he or she knows of any models that you could use as a guide for what you are being asked to do. Such models may be found in closed-case files, manuals, formbooks, and practice texts. They may also be available on the Internet.

Caution is needed whenever using models. Every new legal problem is potentially unique. What will work in one case may not work in another. All model is a guide, a starting point and nothing more. Changes in the model will often be needed in order to adapt it to the particular facts of your assignment. Whenever you find a possible model your supervisor has not seen, bring it to his or her attention and ask whether it can be adapted and used. (See also Exhibit 10-5 in Chapter 10 on how to avoid abusing standard forms.)

#### 9. Do Some Independent Legal Research on the Instructions You Are Given.

Often you will be told what to do without being given more than a cursory explanation of why it needs to be done that way. Most instructions for legal tasks are based upon the requirements of the law. Al complaint, for example, is served on an opposing party in a designated way because the law has imposed rules on how such service is to be made. You may be asked to serve a complaint in a certain way without being told what section of the state code (or of your court rules) requires it to be served in that way. It would be highly impractical to read all the law that is the foundation for an assigned task. It is not necessary to do so and you would not have time to do so.

What you can occasionally do, however, is focus on selected instructions for an assignment and do some background legal research to gain a greater appreciation for why the instructions were necessary. (See the checklist on doing background legal research in Exhibit 11-11 in Chapter 11.) You will probably have to do such legal research on your own time unless the assignment you are given includes doing some legal research. Research can be time-consuming, but you will find it enormously educational. It can place a totally new perspective on the assignment and, indeed, on your entire job.

#### 10. Get on Routing and Mailing Lists for New Legal Material.

Allaw office frequently buys publications for its law library that are relevant to its practice. The publications can include legal treatises and legal periodicals. Before these publications are shelved in the library, they are often routed to the attorneys in the office so that they can become acquainted with current legal writing that will be available in the library. Each attorney usually keeps the publication for a few hours or a few days for review before passing it on to the next person on the mailing list. If the actual publication is not passed around in this manner, those on the mailing list might receive brief summaries of recent publications or photocopies of their tables of contents. Another option in some offices is the use of email to inform attorneys of new publications.

Ask to be included on these routing or mailing lists. The publications are often excellent self-education opportunities, particularly the articles in legal periodicals.

You can also subscribe to free Internet alerts on your area of practice. Findlaw (www.findlaw.com) allows you to subscribe to free newsletters that give summaries of recent cases in designated areas of law (newsletters.findlaw.com). For example, one of the areas covered by the newsletters is family law. If your office practices in this area, you should consider subscribing.

#### 11. Ask Secretaries and Other Paralegals for Help.

Secretaries and paralegals who have worked in the office for a long time can be very helpful to you if you approach them properly. Everybody wants to feel important. Everybody wants to be respected. When someone asks for something in a way that gives the impression he or she is entitled to what is being sought, difficulties usually result. Think of how you would like to be approached if you were in the position of the secretary or paralegal. What behavior or attitude of another employee would irritate you? What would make you want to go out of your way to cooperate with and assist a new employee who needs your help? Your answers (and sensitivity) to questions such as these will go a long way toward enabling you to draw on the experience of others in the office.

#### 12. Obtain Feedback on an Assignment Before the Date It Is Due.

Unless the assignment you are given is a very simple one, do not wait until the date it is due to communicate with your supervisor. If you are having trouble with the assignment, you will want to check with your supervisor as soon as possible and as often as necessary. It would be a mistake, however, to contact the supervisor only when trouble arises. Of course, you want to avoid wasting anyone's time, including your own. You should limit your contacts with a busy supervisor to essential matters. You could take the following approach with your supervisor:

Everything seems to be going fine on the project you gave me. I expect to have it in to you on time. I'm wondering, however, if you could give me a few moments of your time. I want to bring you up to date on where I am so that you can let me know if I am on the right track.

Perhaps this contact could take place on the phone or during a brief office visit. Suppose that you have gone astray on the assignment without knowing it? It is obviously better to discover this before the date the assignment is due. The more communication you have with your supervisor, the more likely it is that you will catch such errors before a great deal of time is wasted.

#### 13. Ask to Participate in Office and Community Training Programs.

Sometimes a law office conducts training sessions for its attorneys. Ask if you can be included. Bar associations and paralegal associations often conduct sessions on legal topics relevant to your work. They are part of what is called *continuing legal education* (CLE). Seek permission to attend some of these sessions if they are held during work hours. If they are conducted after hours, invest some of your own time to attend. Also check into what is available on the Internet. Online CLE has become increasingly popular because you can take courses at any time and in any location where you have access to the Internet. To find out what is available, type "continuing legal education" (or "CLE") in the search boxes of national and local paralegal associations and of bar associations in your area (see Appendix B). Your employer may be willing to pay all or a part of the cost of such courses. Even if you must pay the cost, it will be a worthwhile long-term investment.

#### 14. Ask to Be Evaluated Regularly.

When you first interview for a paralegal position, inquire about the policy of the office on evaluations. Are they conducted on a regular basis? Are they done in writing? Will you know in advance the specific criteria that will be used to evaluate your performance? If you are hired in an office that does

not have a formal evaluation structure or procedure, take the initiative to let supervisors know that you consider evaluations to be important to your professional development. It may take a while for you to feel confident enough to make this known, but the importance of doing so cannot be overestimated.

For a number of reasons, many offices do not have formal evaluations:

- Evaluations can be time-consuming.
- Evaluators are reluctant to say anything negative, especially in writing.
- Most of us do not like to be evaluated: it's threatening to our ego.

Go out of your way to let your supervisor know that you want to be evaluated and that you can handle criticism. If you are defensive when you are criticized, you will find that the evaluations of your performance will go on behind your back! Such a work environment is obviously unhealthy. Consider this approach that a paralegal might take with a supervisor:

I want to know what you think of my work. I want to know where you think I need improvement. That's the only way I'm going to learn. I also want to know when I'm doing things correctly, but I'm mainly interested in your suggestions on what I can do to increase my skills.

If you take this approach and mean it, the chances are good that you will receive constructive criticism and gain a reputation for professionalism.

#### 15. Proceed One Step at a Time.

Perhaps the most important advice you can receive in studying law in school and on the job is to concentrate on what is immediately before you. Proceed one step at a time. What are your responsibilities in the next fifteen minutes? Block everything else out. Make the *now* as productive as you can. Your biggest enemy is worry about the future: worry about the exams ahead of you, worry about your family, worry about the state of the world, worry about finding employment, etc. Leave tomorrow alone! Worrying about it will only interfere with your ability to make the most of what you must do now. Your development in the law will come slowly, in stages. Map out these stages in very small time blocks—beginning with the time that is immediately ahead of you. If you must worry, limit your concern to how to make the next fifteen minutes shine.

#### **SUMMARY**

Legal education is a lifelong endeavor; a competent paralegal never stops learning about the law and the skills of applying it. Al number of important guidelines will help you become a good student in the classroom and on the job. Do not let the media blur your understanding of what the practice of law is actually like. To avoid studying in a vacuum, know the goals of an assignment. Organize your day around a study plan. Assess your study habits, such as how you handle distractions or how you commit things to memory. Then promise yourself that you will do something about your weaknesses.

Increase your proficiency in the basics of writing. How many of the rules about the comma can you identify? Do you know when to use *that* rather than *which* in your sentences? How many of your paragraphs have topic sentences? Are there zero spelling errors on every page of your writing? You have entered a field where the written word is paramount. You must take personal responsibility for the improvement of your grammar, spelling, and composition skills. Use the law library

to help you understand difficult areas of the law. But don't expect absolute clarity all the time. Seek out evaluations of your work. Become a skillful note taker. Get into the habit of looking for definitions.

These suggestions also apply once you are on the job. Don't pretend you understand what you don't. Repeat instructions back to your supervisor before you begin an assignment. Ask for due dates. Ask for priorities if you are given several things to do. Write down your instructions in your own notebook or journal. Become an avid user of checklists, including those you create on your own. Find out if an assignment has been done by others in the past. If so, seek their help. Try to find a model and adapt it as needed. Be prepared to do some independent research. Ask if you can be on internal office mailing lists for new publications. Find out what's available on the Internet to keep current in your area of the law. Participate in training programs at the law office and in the legal community. Ask to be evaluated regularly. Seek feedback before an assignment is due.

#### **REVIEW QUESTIONS**

- 1. Why must a student of law be a perpetual student?
- 2. Distinguish between substantive and procedural law.
- 3. What are the two dimensions of your legal education?
- 4. How do you design a study plan?
- 5. How can you increase your available productive time?
- 6. What are some of the major interferences with effective studying?
- 7. What are some of the techniques of active studying?
- 8. What steps can you take on your own to increase your proficiency in grammar, spelling, and composition?
- What danger do you need to be aware of when using computer spell checkers?
- 10. What are readability formulas and how can they assist you?
- 11. How can study groups help you?
- 12. How can legal research help you handle difficult aspects of any course?
- 13. What is meant by expecting ambiguity in the law?
- 14. Why is note taking important in the study and practice of law?

- 15. What are some of the steps you can take to increase your powers of retention?
- 16. Where and how can you obtain feedback on your schoolwork?
- 17. What is meant by playing "emperor's clothes" with the assignments you receive on the job?
- 18. What are the major techniques of receiving instructions from your supervisor on assignments?
- 19. Why is it important to obtain a specific due date and a statement of priorities for assignments?
- 20. How can you translate instructions into checklists?
- 21. How can you obtain help on an office assignment from practice guides, manuals, models, independent legal research, secretaries, and other paralegals?
- 22. How can you obtain feedback on an assignment from a busy supervisor?
- 23. Why are performance evaluations sometimes ineffective, and what can you do to obtain meaningful ones on the job?

#### **HELPFUL WEBSITES**

#### **Study Techniques**

- www.how-to-study.com
- www.adprima.com/studyout.htm

#### **Note Taking**

- www.sas.calpoly.edu/asc/ssl/notetaking.systems.html
- www.csbsju.edu/academicadvising/help/lec-note.htm
- www.dartmouth.edu/~acskills/success/notes.html
- www.academictips.org/acad/literature/notetaking.html

#### **Online Flashcards for Paralegals**

 www.flashcardexchange.com (type "paralegal" in the search box)

#### **Taking Tests**

- www.testtakingtips.com/test
- www.studygs.net/tsttakl.htm

#### Google, Bing, or Yahoo Searches

(On these search engines, run the following searches.)

- grammar help
- study skills
- how to study
- memory skills
- note taking
- test taking

- supervision skills
- "being supervised"memory skills
- study groups
- study groupson the job learning

#### **ENDNOTES**

<sup>2</sup> JamieCollins, What It Takes to Survive in a Law Office ... and Live to Tell

*the Tale*, KNOW, The Magazine for Paralegals (The Paralegal Society, September 7, 2012) (theparalegalsociety.wordpress.com).

<sup>3</sup> Deborah Bogen, Paralegal Success 79 (2000).

<sup>&</sup>lt;sup>1</sup>Lynda Richardson, *From the Bench, Judgment and Sass*, New York Times, March 27, 2001, at p. A25.







# The Paralegal in the Legal System

- 1 Introduction to the Paralegal Career
  - 2 Paralegal Employment
- 3 On-the-Job Realities: The Assertive Paralegal
  - 4 The Regulation of Paralegals
  - 5 Attorney Ethics and Paralegal Ethics
    - **6** Introduction to the Legal System



### INTRODUCTION TO THE PARALEGAL CAREER

#### CHAPTER OUTLINE

- Launching Your Career
- Do You Know What Time It Is?
- The Scope of Your Legal Education
- Major Players

- Job Titles
- Definitions of a Paralegal
- Paralegal Fees
- Career Ladders in Large Law Offices
- Paralegal Salaries
- Factors Influencing the Growth of Paralegalism
- "Oh, You're Studying Law?"
- Your Online Reputation
- Conclusion

#### CHAPTER OBJECTIVES

#### After completing this chapter, you should be able to:

- Understand the extent of legal disputes in the country.
- Describe the importance of precision in the practice of law.
- List the major organizations in paralegalism and the roles that they play.
- Know the major classifications of paralegal titles.
- Know the definition of a paralegal.
- Outline the major categories of fees.
- Understand the importance of Missouri v. Jenkins.
- Identify the major employers that have career ladders for paralegals.
- Identify the factors that determine a paralegal's salary.
- Understand the factors that have contributed to the development and growth of paralegalism.

#### LAUNCHING YOUR CAREER

Welcome to the field! You probably fall into one or more of the following categories:

- You have never worked in a law office.
- You are or once were employed in a law office and now want to upgrade your skills.
- You want to explore the variety of careers in law for which a paralegal education can be a point of entry.

As you begin your legal education, you may have a large number of questions:

- What is a paralegal?
- Where do paralegals work?
- What are the functions of a paralegal?
- How do I obtain a job?
- How do attorneys and paralegals work together as a team?
- What is the difference between a paralegal and the clerical staff of a law office?
- How do new paralegals adjust to the realities of a law practice?

- How is the paralegal field regulated? Who does the regulating and for what purposes?
- What are the ethical guidelines that govern paralegal conduct?
- What career options are available once a paralegal has obtained experience?

It is an exciting time to be working in the law. So many aspects of our everyday lives have legal dimensions. The law plays a dominant role in a spectrum of issues that range from the status of the unborn to the termination of life-support systems. More than 107 million cases are filed every year in state and federal courts. One study concludes that 52 percent of Americans have a legal problem. One out of three will need the advice of an attorney in the next twelve months.<sup>2</sup> In a large survey of U.S. corporations, 87 percent reported that one or more lawsuits were commenced against them in the past 12 months.<sup>3</sup> The news media are preoccupied with laws that have been broken, laws that no longer make sense, and laws that ought to be passed to create a more just society. Caught up in the whirlwind of these debates is the attorney and the attorney's team, a prominent member of which is the paralegal.

A prominent paralegal was once asked why she became a paralegal. "Deep down," she said, "a good number of us entered the profession because legal proceedings touch everyone on a very personal basis several times during a lifetime, so it is a profession that everyone can relate to. As a paralegal, you . . . also gain the satisfaction of knowing you were able to help solve a problem for a client."3a

There are other satisfactions as well. Once you have completed your paralegal education and gained experience on the job, you can explore a variety of career options within traditional paralegal employment settings. In addition:

- An experienced corporate paralegal in a law firm might leave the firm to take a higherpaying position as a securities analyst for a corporation.
- An experienced estates paralegal at a law firm might leave for a more lucrative position as a trust administrator at a bank.

An extensive list of these law-related jobs is given in Exhibit 2-24 at the end of Chapter 2.

Perhaps the most dramatic example of a paralegal position being used as a stepping-stone for other jobs is the large number of former paralegals who have used their legal training and experience to run for political office, particularly at the city and county levels of government. Peggy Marino, for example, used her paralegal background to help her win a seat on the Board of Education in New Jersey. Peggy now serves on the financial and negotiations committees of that Board. She credits her paralegal experience "as having prepared her for those roles." Similarly, when paralegal Deborah Boe was appointed to the Minnesota Board of Medical Practice, she said that she has "read enough legal documents to know how to read between the lines" and "make my way through the issues" that will come before the Board. Representative Mary Squires of the Georgia Legislature feels the same way. She was a paralegal before she ran for office. Because of a paralegal's preparation and organization skills, she felt equipped for the rigorous life of a legislator. After serving in the House, Mary was elected to the state Senate and is currently a candidate for state insurance commissioner. Of course, not all experienced paralegals go into politics, but the experience of these paralegals demonstrates the value and versatility of legal education.

#### DO YOU KNOW WHAT TIME IT IS?

Perhaps you're wondering what working in the law will be like. One of the hallmarks of legal work is its diversity. In five years, if you meet a fellow classmate and compare notes on what your workdays are like as paralegals, you'll see similarities, but you will probably be startled by the differences.

One way to gauge what working in a law office might be like is to answer a particular question. Stop what you are doing for a moment and answer this question:

Do you know what time it is?

Depending on when you are reading this book, you will probably look at your watch or the clock on the cell phone and answer 9:45 A.M., 1:30 P.M., 3:32 P.M., 11:28 P.M., etc. There could be hundreds of possible answers. But if you answered in this manner, you've made your first mistake in the study and practice of law. Look at the question again. Read it slowly. You were not asked, What time is it? You were asked, Do you know what time it is? There are only two possible answers to this question: yes or no.

Welcome to the law! One of the singular characteristics of the field you are about to enter is its *precision* and *attention to detail*. Vast amounts of time can be wasted by answering the wrong question. In fact, attorneys will tell you that one of the most important skills in the law is the ability to identify and focus on the question—the issue—that must be resolved.

Developing this sensitivity begins with the skill of *listening*—listening very carefully. It also involves *thinking before responding*, noting distinctions, noting what is said and what is not said, being aware of differences in emphasis, and being aware that slight differences in the facts can lead to dramatically different conclusions. Even relatively minor variations in punctuation can sometimes have legal consequences. Compare, for example, the following two statements and note the dramatic difference in meaning that results from the placement of commas:

Johnson said the officer hid the evidence.

Johnson, said the officer, hid the evidence.

In 2008, chief justice of the United States, John Roberts, administered the presidential oath of office by having Barack Obama:

"solemnly swear" that "I will execute the office of President of the United States faithfully."

But the Constitution states that the adverb "faithfully" should go before the verb that it modifies:

"solemnly swear" that "I will faithfully execute the office of President of the United States."

After the inauguration ceremony, the chief justice returned and administered the oath again, this time "by the book."

The hallmark of a professional is precision in language and the ability to identify what is different or unique about a particular person or situation. In this sense, highly competent paralegals are true professionals.

Another major characteristic of law is its focus on definitions and intentions. Attorneys are preoccupied with questions such as "What definition did the legislature intend?" and "What does that clause mean?" For example, if a will says, "I give \$30,000 to my three children, Bill, Grace, and Sam," does each child receive \$30,000 or is this amount to be split among the three children so that each receives \$10,000? The answer depends on the intent of the deceased when the will was written. On September 11, 2001, when two airplanes struck the two towers of the World Trade Center sixteen minutes apart, was the landlord of the towers entitled to one insurance payment of \$3.5 billion or two payments totaling \$7 billion? The answer turned on the definition of the word occurrence in the insurance contracts. Was there one attack, one "occurrence," or two? (After extensive litigation, the landlord was not successful in convincing the courts that there were two occurrences under all of the insurance contracts involved in the case.) Another dramatic example of controversy generated by a definition occurred during a famous grand jury hearing that was videotaped and later played on national television. The witness giving testimony under oath was a graduate of the Yale Law School and a former constitutional law professor. In testimony that captivated the nation, he said that his answer to a particular question (involving sexual relations with an intern) would depend on what the "definition of 'is' is" in the question. The attorney giving this testimony was President Bill Clinton.

In a recent discussion on LinkedIn, the social network for professionals, paralegals were asked about the career they have chosen. Here are some of the responses:

- "I'm constantly intellectually stimulated" and have "made amazing friends."
- "I've been with the same firm for 17 years (longer than any of my romantic relationships!) and still have days when my heart beats faster and am excited about something I am doing, usually digging up hard-to-find information, helping a client get much-needed benefits, learning new technology, or writing substantive content. Each day brings something new."
- "If you let it be known that you are looking for a challenge, you will certainly find it in this career field!"

Not every case in a law office makes the headlines, and not every paralegal is satisfied with all aspects of his or her job. Yet the potential is there for a paralegal career to provide substantial fulfillment and a broad range of opportunities. In short, you have a great deal ahead of you in the study of law; it's going to be a fascinating adventure.

#### THE SCOPE OF YOUR LEGAL EDUCATION

Legal education is a very broad area of study. You will be covering a great deal of material in your courses. Will you use all of this material on the job? Not necessarily. Yet the material must be covered in order to comprehend the legal system and the world of attorneys. For example, you will learning about appellate briefs and sophisticated legal research even though many paralegals never draft such briefs nor perform such research. Yet you need to have a basic understanding of such tasks in order to communicate with attorneys and to perform assignments related to these tasks. In short, understanding the role of a paralegal requires an understanding of and exposure to a wide range of paralegal and attorney skills as you begin the journey of comprehending the practice of law and the legal system as a whole.

Areas that are essential to the performance of specific paralegal tasks include:

- the functions of courts, legislatures, and administrative agencies
- the rules of professional responsibility (ethics) governing attorneys and paralegals
- the rules of grammar and clear communication
- the ability to state facts accurately and comprehensively
- the basics of computer hardware and software on your desk and "in the cloud"
- the ability to adapt to the different standards of different attorneys in an environment where time pressures can be intense

Without a grasp of these essentials, a paralegal career can flounder. Hence we will spend a good deal of time on them. Yet we will also cover the big picture, which includes tasks that many attorneys never assign paralegals to perform. This broader perspective is needed to develop into a well-rounded paralegal and to be ready for the great diversity that awaits you within the legal community.

#### MAJOR PLAYERS

During this course, we will meet many organizations. Here is an overview of some of the major players (not necessarily listed in order of influence):

- National Federation of Paralegal Associations (NFPA)
- National Association of Legal Assistants (NALA)
- NALS, The Association for Legal Professionals (NALS)
- American Bar Association Standing Committee on Paralegals (SCOP)
- Your state bar association (see Appendix B)
- Your local paralegal association (see Appendix B)

These organizations are covered in some detail throughout the remaining chapters of the book, but a brief word about each will be helpful at this point.

All six organizations extensively promote (and most offer) continuing legal education (CLE), which is training in the law (usually short term) that a person receives after completing his or her formal legal training or after becoming employed.

#### NATIONAL FEDERATION OF PARALEGAL **ASSOCIATIONS (NFPA)**

(www.paralegals.org)

The NFPA is an association of more than fifty state and local paralegal associations throughout the country. (See Appendix B.) Individual paralegals can also be members, but most are connected to NFPA through their local paralegal association. More than 10,000 paralegals are part of the NFPA network. NFPA offers voluntary certification for entry-level paralegals (PCCE: the Paralegal CORE Competency Exam). By entrylevel, we mean that the certification credential can be earned without having paralegal job experience. (In the job market, entry-level also refers to a starting position for someone with little or no experience.) NFPA has another voluntary certification program for paralegals with experience (PACE: the Paralegal Advanced Competency Exam). We discuss these credentials in Chapter 4.

#### continuing legal education

(CLE) Training in the law (usually short term) that a person receives after completing his or her formal legal training or after becoming employed.

certification The process by which a nongovernmental organization grants recognition to a person who has met the qualifications set by that organization.

# NATIONAL ASSOCIATION OF LEGAL ASSISTANTS (NALA)

(www.nala.org)

NALA is primarily an association of individual paralegals, although eighty-one state and local paralegal associations are affiliated with NALA. (See Appendix B.) More than 18,000 paralegals in the country are represented by NALA. It offers voluntary certification for entry-level paralegals (CLA, Certified Legal Assistant/Certified Paralegal) and for paralegals with experience (APC, Advanced Paralegal Certification). See Chapter 4.

#### NALS, THE ASSOCIATION FOR LEGAL PROFESSIONALS

(www.nals.org)

The oldest national association in the country is NALS, the Association for Legal Professionals. (See Appendix B.) The association consists of more than 4,000 paralegals and legal secretaries in 100+ chapters throughout the country. NALS offers a voluntary certification examination (PP, Professional Paralegal). See Chapter 4. The National Association of Legal Assistants (NALA) was initially formed as a breakaway association from NALS. Both associations are still headquartered in Tulsa, Oklahoma. (At one time, NALS stood for National Association of Legal Secretaries; the organization kept the abbreviation but changed its meaning.)

See Exhibit 1-1 concerning convention activity of NFPA, NALA, and NALS.

#### Exhibit 1-1

#### Annual Conventions: NFPA, NALA, and NALS



Annual convention of the National Federation of Paralegal Associations (NFPA).



Annual convention of the National Association of Legal Assistants (NALA).



Continuing legal education (CLE) session of NALS, the Association for Legal Professionals.

Reprinted with permission of the National Federation of Paralegal Associations; photo courtesy of NALA and Ken Frakes; reprinted by permission of Nancy Harris (PP, PLS) and NALS.

# AMERICAN BAR ASSOCIATION STANDING COMMITTEE ON PARALEGALS (SCOP)

(www.americanbar.org/groups/paralegals.html)

The American Bar Association (ABA) is a voluntary association of attorneys; no attorney must be a member. Yet it is a powerful organization because of its resources, its prestige, and the large number of attorneys who have joined. The ABA has a Standing Committee on Paralegals (SCOP) that has had a significant impact on the growth of the field. The role of the ABA has included the publication of research on paralegals, the establishment of a voluntary program of approving paralegal schools, and the creation of an associate membership category for paralegals.

#### STATE BAR ASSOCIATION OF YOUR STATE

(see Appendix B)

Most state bar associations have been active regarding the role of paralegals in the practice of law. Many of the associations have issued (1) guidelines on the proper use of paralegals by attorneys and (2) ethical opinions that apply the state's ethics code to an attorney's use of paralegals. We examine such guidelines and opinions in Chapter 5 and in Appendix D. Some state bar associations have followed the lead of the ABA and allow paralegals to become associate or affiliate members.

#### YOUR LOCAL PARALEGAL ASSOCIATION

(see Appendix B)

Paralegal associations fall into at least fourteen categories, some of which overlap:

- 1. Nationwide (e.g., NFPA, NALA, NALS)
- 2. Statewide (e.g., Illinois Paralegal Association)
- 3. Regionwide encompassing more than one state (e.g., Rocky Mountain Paralegal Association)
- 4. Regionwide within a state (e.g., South Florida Paralegal Association)
- 5. Countywide (e.g., Santa Clara County Paralegal Association)
- 6. Citywide (e.g., New York City Paralegal Association)
- 7. Practice specific (e.g., Houston Corporate Paralegal Association)
- 8. Division of a bar association (e.g., Paralegal Division of the State Bar of Texas)
- 9. Association of associations (e.g., Empire State Alliance of Paralegal Associations)
- 10. Schoolwide (e.g., Fresno City College Paralegal Association)
- 11. Manager-focused (e.g., International Practice Management Association)
- 12. Association whose membership is limited to paralegals (e.g., Orange County Paralegal Association)
- 13. Association whose membership consists of paralegals, legal secretaries, and other nonattorneys (e.g., NALS)
- 14. Foreign (e.g., Canadian Paralegal Association)

Most paralegals in the country have joined a state, regional, county, city, or bar-affiliated paralegal association. (They are collectively referred to in this book as *local paralegal associations*.) Through CLE and **networking**, a great many paralegals have found major career support and inspiration by active participation in their local paralegal association. If any of them have a student membership rate, you should consider joining now, while you are in this course.

#### **OTHER IMPORTANT ORGANIZATIONS**

Although NFPA, NALA, NALS, SCOP, your state bar association, and your local paralegal association will dominate our discussion of paralegalism, we will also be referring to other important groups, such as the following:

- International Practice Management Association (IPMA): an association of paralegal supervisors in large law offices (www.paralegalmanagement.org)
- American Association for Paralegal Education (AAfPE): an association of paralegal schools (www.aafpe.org)
- Association of Legal Administrators (ALA): an association of individuals (mostly nonattorneys) who help administer or manage law offices (www.alanet.org)
- American Alliance of Paralegals, Inc. (AAPI): a national association of paralegals that offers certification without an examination (www.aapipara.org)

networking Establishing contacts and sharing information with people (a) who might become personal or professional resources for you and (b) for whom you might become a personal or professional resource.

#### **JOB TITLES**

#### PARALEGAL AND LEGAL ASSISTANT

For convenience, this book uses the title *paralegal*. At one time, another common title was *legal assistant*. For years most people agreed that these titles were synonymous, just as the titles *attorney* and *lawyer* are synonymous. Yet the titles *paralegal* and *legal assistant* have been the source of confusion and debate. For example, in some state governments, legal-assistant

positions are attorney positions. In the federal government, most legal secretaries are called legal assistants. More significantly, many private law offices have changed the title of their legal secretaries to *legal assistants*| without changing the requirements for their position or the duties they perform. Those who favor this title change say that it reflects the broader role that many secretaries perform. Cynics, however, argue that the change is due to the fact that courts can give attorneys separate fees for the work of their legal assistants but not for that of their secretaries. We will be examining such fees later in this chapter.

The shift in title from *secretary* to *legal assistant*| has not been universal; there are still individuals called legal secretaries in offices throughout the country. Yet the number of legal secretaries now called legal assistants is substantial, and this has led many to conclude that the legal assistant and paralegal titles are no longer synonymous. There is an emerging preference for the *paralegal*| title. For example, many local associations have voted to change their name from legal-assistant association to paralegal association. One of the largest paralegal organizations in the country, the Legal Assistant Division of the State Bar of Texas, changed its name to the Paralegal Division of the State Bar of Texas. As indicated, an important committee of the American Bar Association is the Standing Committee on Paralegals. Its former name was the Standing Committee on Legal Assistants.

The National Association of Legal Assistants (NALA), however, is adamant that "the terms legal assistant and paralegal are used interchangeably." This is also the view of the International Practice Management Association (IPMA), yet even IPMA has gone through a name change—it was once known as the Legal Assistant Management Association. NALA itself has acknowledged this trend. As we will see in Chapter 4, if you take and pass NALA's certification exam, you have a choice of being called a certified legal assistant (CLA) or a certified paralegal (CP). The former title has been available since the 1970s; the latter was added as an option in 2004 and is now preferred by those who pass the exam. It is true that some laws (discussed in Chapter 4) treat paralegals and legal assistants as synonyms. Yet the momentum continues on many fronts to separate the positions of legal assistant and paralegal and to prefer the latter.

To summarize:

- At one time, everyone considered paralegal and legal assistant to be interchangeable titles; they were synonymous.
- Paralegal became the preferred title after many attorneys allowed their secretaries to be called legal assistants.
- Some laws and some organizations, however, still maintain that the titles are interchangeable.

#### **OTHER TITLES**

This is not to say that *paralegal* and *legal assistant* are the only titles in the field. Far from it. Many other titles are in use. Several reasons account for this diversity.

First, there are no national standards regulating the paralegal field. Every state is free to regulate or to refuse to regulate a particular occupation. Second, most states have not imposed the kind of regulation on paralegals that would lead to greater consistency on titles within a particular state. For example, most states have not imposed minimum educational requirements or licensing. Hence, few restrictions exist on who can call themselves paralegals, legal assistants, or related titles. Although there are some major exceptions in states such as California, Florida, South Dakota, and Maine that limit who can use specific titles (discussed in Chapter 4), for the most part there are no such restrictions.

To begin sorting through the maze, we examine two broad categories of workers: those who are employees of attorneys (often called **traditional paralegals**) and those who are **independent contractors**. In general, independent contractors are self-employed persons who operate their own business and contract to do work for others. In general, those who hire the services of independent contractors do not control many of the administrative *details* of how the services are performed. The *objectives* or end products of the services, however, are controlled by those who hire them.

We will be examining two kinds of independent contractors in the legal field: those who sell their services to attorneys (a relatively small group) and those who sell their services to the public without attorney supervision (a larger and more controversial group). Within these two categories, you will find considerable diversity (and overlap) in the titles that are used.

# traditional paralegals A paralegal who is an employee of an attorney.

# independent contractors A self-employed person who operates his or her own business and contracts to perform tasks for others. In general, the latter do not control many of the administrative details of how the work is performed.